

Economic Research

Structural Error Correction Models : Instrumental Variables Method and an Application to an Exchange Rate Model

Jaebeom Kim, Masao Ogaki, Minseok Young

Working Paper No. 502 October 2003

UNIVERSITY OF **ROCHESTER**

Structural Error Correction Models: Instrumental Variables Methods and an Application to an Exchange Rate Model

Jaebeom Kim University of St. Thomas

Masao Ogaki^{*} The Ohio State University

Minseok Yang Seoul Cyber University

October, 2003

Abstract

Error correction models are widely used to estimate dynamic cointegrated systems. In most applications, estimated error correction models are reduced form models. As a result, nonstructural speed of adjustment coefficients are estimated in these applications. A single equation instrumental variable method can be used to estimate a structural speed of adjustment coefficient. This paper develops a system instrumental variable method to estimate the structural speed of adjustment coefficient in an error correction model. This method utilizes Hansen and Sargent's (1982) instrumental variable estimator for linear rational expectations models, and is applied to an exchange rate model with sticky prices.

^{*}We thank seminar participants at the Bank of Japan, International Monetary Fund, the Ohio State University, Southern Methodist University, University of Houston, University of Memphis, University of Michigan, the 1997 North American Winter Meeting of the Econometric Society, and the 2003 Konstanz Seminar for Monetary Theory and Policy for their comments. Special thanks are due to Steve Cecchetti, Boris Hoffman, G.S. Maddala, Nelson Mark, Mat Shapiro, Bill Smith, Alan Viard, and Charles Whiteman.

1 Introduction

Davidson, Hendry, Srba, and Yeo's (1978) Error Correction Model (ECM) is widely used to estimate dynamic cointegrated systems.¹ As the Granger Representation Theorem shows (see Engle and Granger (1988)), an ECM representation exists when the variables are cointegrated and vice versa.² The standard ECMs are reduced form models just as VAR models are as pointed out by Urbain (1992) and Boswijk (1994,1995). As in VAR models, identification of structural shocks is an important issue for structural ECMs. King, Plosser, Stock, and Watson (1991), Jang (2000), and Jang and Ogaki (2003) develop methods to identify structural shocks with short-run and long-run restrictions.³

Another important issue for the structural ECM is estimation of the speed of adjustment toward the long-run equilibrium level. This paper focuses on this issue. In a structural ECM, at least one linear combination of variables gradually adjusts to the long-run equilibrium level with a constant speed of adjustment. In general, the speed of adjustment coefficient in a structural ECM is different from the speed of adjustment coefficient in its reduced form ECM. As an example, we will show that they are different in an exchange rate model with sticky prices. In the exchange rate model, the structural speed of adjustment coefficient is not a deep structural parameter, but it is equal to one minus the first order autoregressive coefficient for the log real exchange rate.⁴ Hence the structural speed of adjustment coefficient can be used to compute the half-life of the real exchange rate, and is a parameter of interest. However, the reduced form speed of adjustment is a nonlinear function of the structural speed of adjustment and the interest elasticity of money demand. Hence the reduced form speed of adjustment coefficient in the ECM cannot be directly compared with the half-life estimates of real exchange rates in the literature (see, e.g., Rogoff (1996) for a survey, and Kilian and Zha (2002) and Murray and Papell (2002) for more recent works).

Standard estimation methods for ECMs such as Engle and Granger's two step method and Johansen's (1988) Maximum Likelihood method estimate the reduced form speed of adjustment

¹An alternative method is levels VAR without imposing unit roots. Estimators which are based on a levels VAR are more robust but are usually less efficient than those based on an ECM if the restrictions regarding nonstationarity and cointegration imposed by the ECM are true.

²The theorem should be used with caution because there exist economic models in which the regularity conditions of the theorem do not apply as shown in Ogaki (1998). However, the model in this paper is subject to this criticism. ³The word Vector ECM (VECM) is usually used when identification for structural shocks is studied.

⁴As explained later, the coefficient cannot be estimated by Ordinary Least Squares with measurement errors. However, the structural coefficient is equal to one minus the first order autoregressive coefficient of the true value of

coefficient rather than the structural speed of adjustment coefficient. A single equation instrumental variable (IV) method can be directly applied to a gradual adjustment equation. The main purpose of this paper is to develop a system method that combines the single equation method with Hansen and Sargent's (1982) IV method for linear rational expectations models.

In the single equation method, an IV method is applied to a gradual adjustment equation that describes how a variable slowly adjusts to the long-run equilibrium level in the structural ECM. The system method combines the single equation method with Hansen and Sargent's (1982) method which applies Hansen's (1982) Generalized Method of Moments (GMM) to linear rational expectations models. The system method is more efficient than the single-equation method when the restrictions implied by linear rational expectations models are true. On the other hand, the single equation method is robust to misspecification in the other equations of the structural ECM. Therefore, we can form a test statistic of the restrictions by comparing the results from the two methods.

These methods are applied to an exchange rate model with sticky prices. The model is a onegood version of Mussa's (1982) model, which may be viewed as a stochastic discrete time version of Dornbush's (1976) model. This model includes a gradual adjustment equation, in which the domestic price adjusts to the long run equilibrium level determined by Purchasing Power Parity (PPP) with rational expectations. We refer the speed of adjustment coefficient for this equation as the structural speed of adjustment coefficient. Because the basic idea of the ECM is that variables adjust to their long run levels, it is of interest to examine whether or not the standard estimation methods of the ECM can be used to estimate the structural speed of adjustment coefficient. We will show that, in the exchange rate model, the standard ECM estimation methods do not recover the structural speed of adjustment coefficient.

Data are for the exchange rates of currencies of Canada, France, Germany, Italy, Japan, the United Kingdom, and the United States. Using the single equation method, we obtain positive estimates for the structural speed of adjustment coefficient in most cases.

We then apply the system method to the same data set. In this case the speed of adjustment coefficient can be estimated from the gradual adjustment equation for the domestic price and the rational expectations equation for the exchange rate. We form a specification test by comparing the estimates for the speed of adjustment coefficient from these two equations.

Structural ECMs have been considered by several authors. Urbain (1992) investigates sufficient

conditions for weak exogeneity for structural ECMs that are similar to ours. Boswijk (1994,1995) and Hsiao (1997) discuss the relationship between the ECM and structural simultaneous equations models. However, unlike Urbain and Hsiao, we do not assume that exogenous variables are observed by the econometrician. In our empirical application, it is not natural to assume exogeneity of any variable in the cointegrated system. Papell (1995) derives a reduced form ECM from an exchange rate model that is similar to ours. However, the real exchange rate is nonstationary in his model unlike ours. He applies Phillips' (1991) ML estimator to the reduced form ECM. Dolado, Galbraith and Banerjee (1991) and Gregory, Pagan, and Smith (1993) derive structural ECMs from linear quadratic models. They discuss the difficulties associated with the application of standard estimation methods such as Engle and Granger's (1987) two-step method and Johansen's (1988, 1991) Maximum Likelihood (ML) method to the ECM. They do not combine their method with Hansen and Sargent's (1982) IV method for linear rational expectations models.

The rest of this paper is organized as follows. In Section 2, a structural ECM is presented and its relationship to a reduced form ECM is discussed. Section 3 discusses the single equation and system methods for the structural ECM. Section 4 presents an exchange rate model in which the domestic price slowly adjusts toward the Purchasing Power Parity (PPP) level. In Section 5, the model of Section 4 is augmented to include measurement errors. Section 6 presents our empirical results for the system methods. Section 7 contains concluding remarks.

2 Structural Models and Error Correction Models

In this section, we discuss the relationship between structural models and ECMs. Let \mathbf{y}_t be an *n*-dimensional vector of first difference stationary random variables. We assume that there exist ρ linearly independent cointegrating vectors, so that $\mathbf{A}'\mathbf{y}_t$ is stationary, where \mathbf{A}' is a ($\rho \times$ n) matrix of real numbers whose rows are linearly independent cointegrating vectors. Consider a standard ECM

$$\Delta \mathbf{y}_{t+1} = \mathbf{k} + \mathbf{G}\mathbf{A}'\mathbf{y}_t + \mathbf{F}_1 \Delta \mathbf{y}_t + \mathbf{F}_2 \Delta \mathbf{y}_{t-1} + \dots + \mathbf{F}_p \mathbf{y}_{t-p+1} + \mathbf{v}_{t+1}$$
(1)

where \mathbf{k} (n × 1) vector, \mathbf{G} is a (n × ρ) matrix of real numbers, \mathbf{v}_t is a stationary *n*-dimensional vector of random variables with $\widehat{E}[\mathbf{v}_{t+1}|H_{t-\tau}] = 0$. In many applications $\tau = 0$, but we will give examples of applications in which $\tau > 0.5$ There exist many ways to estimate (1). For example,

⁵We will treat more general cases in which the expectation of ν_{t+1} conditional on the economic agents' information is not zero, but the linear projection of ν_{t+1} onto an econometrician's information set (which is smaller than the

Engle and Granger's two step method or Johansen's Maximum Likelihood methods can be used.

Many applications of standard ECMs give elements in **G** structural interpretations as parameters of the speed of adjustment toward the long-run equilibrium represented by $\mathbf{A}'\mathbf{y}_t$. It is of interest to study conditions under which the elements in **G** can be given such a structural interpretation. In the model of Section 4, the domestic price level gradually adjusts to its PPP level with a speed of adjustment parameter *b*. We will investigate conditions under which *b* can be estimated as an element in **G** from (1).

In most applications, (1) is a reduced form model. A class of structural models can be written in the following form of a structural ECM:

$$\mathbf{C}_{0} \Delta \mathbf{y}_{t+1} = \mathbf{d} + \mathbf{B} \mathbf{A}' \mathbf{y}_{t} + \mathbf{C}_{1} \Delta \mathbf{y}_{t} + \mathbf{C}_{2} \Delta \mathbf{y}_{t-1} + \dots + \mathbf{C}_{p} \mathbf{y}_{t-p+1} + \mathbf{u}_{t+1}$$
(2)

where \mathbf{C}_i is a $(\mathbf{n} \times \mathbf{n})$ matrix, \mathbf{d} is an $(\mathbf{n} \times 1)$ vector, and \mathbf{B} is an $(\mathbf{n} \times \rho)$ matrix of real numbers.⁶ Here \mathbf{C}_0 is a nonsingular matrix of real numbers with ones along its principal diagonal, \mathbf{u}_t is a stationary *n*-dimensional vector of random variables with $\widehat{E}[\mathbf{u}_{t+1}|H_{t-\tau}] = 0$. Even though cointegrating vectors are not unique, we assume that there is a normalization that uniquely determines \mathbf{A} so that parameters in \mathbf{B} have structural meanings. For the rest of this paper, we assume that the gradual adjustment toward the long-run equilibrium in the first equation in Equation (2) is of particular interest to the researcher. Therefore, the first row of \mathbf{B} gives the structural parameters of interest.

In order to see the relationship between the standard ECM and the structural ECM, we premultiply both sides of (2) by \mathbf{C}_0^{-1} to obtain the standard ECM (1), where $\mathbf{k} = \mathbf{C}_0^{-1}\mathbf{d}$, $\mathbf{G} = \mathbf{C}_0^{-1}\mathbf{B}$, \mathbf{F}_i $= \mathbf{C}_0^{-1}\mathbf{C}_i$, and $\boldsymbol{\nu}_t = \mathbf{C}_0^{-1}\mathbf{u}_t$. Thus the standard ECM estimated by Engle and Granger's two step method or Johansen's Maximum Likelihood methods is a reduced form model. Hence it cannot be used to recover structural parameters in \mathbf{B} , nor can the impulse-response functions based on $\boldsymbol{\nu}_t$ be interpreted in a structural way unless some restrictions are imposed on \mathbf{C}_0 .

As in a VAR, various restrictions are possible for C_0 . One example is to assume that C_0 is lower triangular. If C_0 is lower triangular, then the first row of **G** is equal to the first row of **B**, and structural parameters in the first row of **B** are estimated by the standard methods used to estimate an ECM. In the exchange rate model in this paper, the restriction that C_0 is lower triangular is not

economic agents' information set) is zero.

⁶If the deterministic cointegration restriction (see Ogaki and Park, 1998, for this terminology) is not satisfied, then a linear trend term needs to be added to Equation (2).

attractive. However, as we discuss in Section 4.3, the structural ECM from the one-good version of the exchange rate model does not satisfy the restriction that C_0 is lower triangular for any ordering of the variables. Even though some structural models may be written in lower triangular form, our example suggests that many structural models cannot be written in that particular form.

3 The Instrumental Variables Methods

Because standard methods of estimating reduced form ECMs may not recover the structural parameters of interest in **B**, we consider two instrumental variables methods. The single equation method simply applies an IV estimator to a gradual adjustment equation. The system method combines the single equation method with Hansen-Sargent IV estimator. These methods do not require restrictions on C_0 .

3.1 The Single Equation Method

First, we consider a single equation method, which applies an IV method to a gradual adjustment equation. Imagine that we are interested in estimating the first row of Equation (2). In some applications, the cointegrating vectors are known, and thus the values of \mathbf{A} are known. It should be noted that ordinary least squares may be applicable in this case of known cointegrating vectors.⁷ In other applications, the values of \mathbf{A} are unknown. In the case of the unknown cointegrating vectors, a two-step method that is similar to Engle and Granger's (1987) and Cooley and Ogaki's (1996) methods can be used. In this two-step method, the cointegrating vectors are estimated in the first step.

In the first step, we estimate \mathbf{A} , using a method to consistently estimate cointegrating vectors. There exist many methods to estimate cointegrating vectors. Johansen's Maximum Likelihood (ML) Estimators for Equation (1) can be used for this purpose. If ρ is equal to one, estimators based on regressions that are as efficient as Johansen's ML estimators such as Phillips and Hansen's Fully Modified Estimation Method (1990), Park's (1992) Canonical Cointegrating Regression, and Stock and Watson's (1993) estimators can be used. Ordinary Least Squares estimators are also consistent when ρ is equal to one, but not as efficient as these estimators. We assume that \mathbf{A}_T is the first step estimator, where T is the sample size, and \mathbf{A}_T converges to \mathbf{A} at a faster rate than

⁷In our exchange rate model without measurement errors, ordinary least squres can be applied to an autoregressive regression for the real exchange rate to estimate the structural speed of adjustment coefficient.

 $T^{1/2}.^{8}$

In the single equation method, an IV method is applied to

$$\Delta y_{1,t+1} = d_1 - c_{02}^1 \Delta y_{2,t+1} - \dots - c_{0n}^1 \Delta y_{n,t+1} + \mathbf{b}_1 \mathbf{A'y}_t + \mathbf{c}_1^1 \Delta \mathbf{y}_t + \mathbf{c}_2^1 \Delta \mathbf{y}_{t-1} + \dots + \mathbf{c}_p^1 \Delta \mathbf{y}_{t-p+1} + u_{1,t+1}$$
(3)

where $y_{i,t}$ is the *i*-th element of \mathbf{y}_t , d_1 is the first element in \mathbf{d} , c_{0i}^1 is the *i*-th element of the first row of \mathbf{C}_0 , \mathbf{b}_1 is the first row of \mathbf{B} , \mathbf{c}_i^1 is the first row of \mathbf{C}_i , and $u_{1,t}$ is the first element of \mathbf{u}_t . When $\widehat{E}[u_{1,t+1}|H_{t-\tau}] = 0$, any stationary variable in the information set available at time $t - \tau$ that is correlated with variables in the right hand side of Equation (3) can be used as an instrumental variable. In the case of the known cointegrating vectors, the known values of \mathbf{A} are used in (3). In the case of the unknown cointegrating vectors, \mathbf{A}_T obtained in the first step replaces \mathbf{A} in Equation (3). Because \mathbf{A}_T converges to \mathbf{A} at a faster rate than $T^{1/2}$, the first step estimation does not affect the asymptotic distributions of the second step estimator under regularity conditions.

3.2 The System Method

In this section, we propose an econometric method that combines our single equation method with Hansen and Sargent's (1982) procedure to impose nonlinear restrictions implied by rational expectations models.

Let $\mathbf{y}_t = (y_{1,t}, y_{2,t}, y_{3,t}, y_{4,t})'$ be 4×1 vector of random variables with a structural ECM representation (2). Assume that there exists only one linearly independent cointegrating vector \mathbf{A} such that $\mathbf{A}'\mathbf{y}_t$ is stationary. In the following, \mathbf{y}_t is partitioned into four subvectors, and each subvector is given a different role. For expositional simplicity, we assume that each subvector is one dimensional so that \mathbf{y}_t is a 4×1 vector, and that \mathbf{y}_t has only one cointegrating vector.

The first element of \mathbf{y}_t represents a gradual adjustment as in Equation (3), with nonzero \mathbf{b}_1 where $\widehat{E}[u_{1,t+1}|H_{t-\tau}] = 0$. We assume that the second element of \mathbf{y}_t is related to a discounted sum of expected future values of the fourth element in the following form:

$$\Delta y_{2,t+1} = d_2 - c_{01}^2 \Delta y_{1,t+1} - c_{03}^2 \Delta y_{3,t+1} - c_{04}^2 \Delta y_{4,t+1}$$

$$+ \alpha E[\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1} | I_t] + \varepsilon_{e,t+1}$$
(4)

⁸Usually, \mathbf{A}_T converges at the rate of T, but there are cases where \mathbf{A}_T converges at the rate of $T^{2/3}$ (see West, 1988).

where δ is a positive constant that is smaller than one, and α is a constant. As pointed out by Hansen and Sargent, many linear rational expectations models imply that one variable is a geometrically declining weighted sum of expected future values of other variables.

Hansen and Sargent's (1982) methodology is to project the conditional expectation of the discounted sum, $\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1}$, onto an information set H_t , which is a subset of I_t , the economic agents' information set. Let $\widehat{E}[\cdot|H_t]$ be the linear projection operator conditional on an information set H_t which is a subset of I_t . Replacing the conditional expectation by the linear projection gives

$$\Delta y_{2,t+1} = d_2 - c_{01}^2 \Delta y_{1,t+1} - c_{03}^2 \Delta y_{3,t+1} - c_{04}^2 \Delta y_{4,t+1}$$

$$+ \alpha \widehat{E} [\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1} | H_t] + u_{2,t+1}$$
(5)

where

$$u_{2,t+1} = \varepsilon_{e,t+1} + E[\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1} | I_t] - \widehat{E}[\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1} | H_t]$$
(6)

Because H_t is a subset of I_t , we obtain $\widehat{E}[u_{2,t+1}|H_t] = 0$.

The current and past values of the first difference of the third element of \mathbf{y}_t are used to form the econometrician's information set H_t . Since $\widehat{E}[\cdot|H_t]$ is the linear projection operator onto H_t , there exist possibly infinite order lag polynomials $\beta(L)$, $\gamma(L)$, and $\xi(L)$, such that

$$\widehat{E}[\Delta y_{3,t+1}|H_t] = \beta(L)\Delta y_{3,t} \tag{7}$$

$$\widehat{E}[\Delta y_{4,t+1}|H_t] = \gamma(L)\Delta y_{3,t} \tag{8}$$

$$\widehat{E}\left[\sum_{j=0}^{\infty} \delta^j \Delta y_{4,t+j+1} | H_t\right] = \xi(L) \Delta y_{3,t} \tag{9}$$

Then following Hansen and Sargent (1980, Appendix A), we obtain the restrictions imposed on $\xi(L)$:

$$\xi(L) = \frac{\gamma(L) - \delta L^{-1} \gamma(\delta) [1 - \delta \beta(\delta)]^{-1} [1 - L\beta(L)]}{1 - \delta L^{-1}}$$
(10)

Substituting (9) into (5) gives the equation

$$\Delta y_{2,t+1} = d_2 - c_{01}^2 \Delta y_{1,t+1} - c_{03}^2 \Delta y_{3,t+1} - c_{04}^2 \Delta y_{4,t+1}$$

$$+ \alpha \xi(L) \Delta y_{3,t} + u_{2,t+1}$$
(11)

where $\xi(L)$ is given by (10). We now make an additional assumption that the lag polynomials $\beta(L)$ and $\gamma(L)$ are finite order polynomials, so that

$$\Delta y_{3,t+1} = \beta_1 \Delta y_{3,t} + \beta_2 \Delta y_{3,t-1} + \dots + \beta_p \Delta y_{3,t-p+1} + u_{3,t+1}$$
(12)

$$\Delta y_{4,t+1} = \gamma_1 \Delta y_{3,t} + \gamma_2 \Delta y_{3,t-1} + \dots + \gamma_{p-1} \Delta y_{3,t-p+2} + u_{4,t+1}$$
(13)

where $\widehat{E}[u_{i,t+1}|H_t] = 0$ for i = 3, 4. Here we assume $\beta(L)$ is of order p and $\gamma(L)$ is of order p-1in order to simplify the exposition, but we do not lose generality because any of β_i and γ_i can be zero. Then as in Hansen and Sargent (1982), (10) implies

$$\xi_0 = \gamma(\delta)[1 - \delta\beta(\delta)]^{-1}$$

$$\xi_j = \delta\gamma(\delta)[1 - \delta\beta(\delta)]^{-1}(\beta_{j+1} + \delta\beta_{j+1} + \dots + \delta^{p-j}\beta_p) + (\gamma_j + \delta\gamma_j + \dots + \delta^{p-j}\gamma_p)$$
(14)

for j = 1, ..., p.

In the SECM form (2), we have $\mathbf{B} = [-b, 0, 0, 0]', \mathbf{A} = [1, -1, -1, 0]'$,

$$\mathbf{C}_{0} = \begin{bmatrix} 1 & c_{02}^{1} & c_{03}^{1} & c_{04}^{1} \\ c_{01}^{2} & 1 & c_{03}^{2} & c_{04}^{2} \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix},$$
(15)

and

$$\mathbf{C}_{j} = \begin{bmatrix} c_{j1}^{1} & c_{j2}^{1} & c_{j3}^{1} & c_{04}^{1} \\ 0 & 0 & \alpha\xi_{j} & 0 \\ 0 & 0 & \beta_{j} & 0 \\ 0 & 0 & \gamma_{j} & 0 \end{bmatrix}$$
(16)

for j = 1, ..., p, where $\gamma_p = 0$.

We have now obtained a system of four equations that consist of (3), (11), (12), and (13). Because $E[u_{i,t}|I_{t-\tau}] = 0$ and $\widehat{E}[u_{i,t}|H_t] = 0$, we can obtain a vector of instrumental variables $\mathbf{z}_{1,t}$ in $I_{t-\tau}$ for $\mathbf{u}_{1,t}$ and $\mathbf{z}_{i,t}$ in H_t for $\mathbf{u}_{i,t}$ (i = 2, 3, 4).

Because the speed of adjustment b for $y_{1,t}$ affects the dynamics of other variables,⁹ there will be cross-equation restrictions involving b in many applications in addition to the restrictions in (14). Using the moment conditions $E[\mathbf{z}_{i,t}\mathbf{u}_{i,t}] = 0$ for i=1,...,4, we form a GMM estimator, imposing the restrictions in (14) and the other cross-equation restrictions implied by the model.

⁹Note that only $y_{1,t}$ adjusts slowly, but *b* affects dynamics of other variables because of interactions between $y_{1,t}$ and those variables.

Given estimates of cointegrating vectors from the first step, this system method provides more efficient estimators than the single equation two step method proposed in previous section as long as the restrictions implied by the model are true.¹⁰ On the other hand, the single equation two step method estimators are more robust because misspecification in other equations does not affect their consistency. The cross-equation restrictions can be tested by Wald, Likelihood Ratio type, and Lagrange Multiplier tests in the GMM framework (see, e.g., Ogaki 1993a). When restrictions are nonlinear, Likelihood Ratio type and Lagrange Multiplier tests are known to be more reliable than Wald tests.

4 An Exchange Rate Model with Sticky Prices

In this section, we present a simple exchange rate model in which the domestic price adjusts slowly toward the long-run equilibrium level implied by Purchasing Power Parity (PPP). This model is used to motivate a particular form of a structural ECM in the previous section. The model's two main components are a gradual adjustment equation and a rational expectations equation for the exchange rate. The single equation method in Section 3 is based only on the gradual adjustment equation. The system method utilizes both the gradual adjustment and rational expectations equations.

4.1 The Gradual Adjustment Equation

Let p_t be the log domestic price level, p_t^* be the log foreign price level, and e_t be the log nominal exchange rate (the price of one unit of the foreign currency in terms of the domestic currency). We assume that these variables are first difference stationary. We also assume that PPP holds in the long run, so that the real exchange rate, $p_t - p_t^* - e_t$, is stationary, or $\mathbf{y}_t = (p_t, e_t, p_t^*)'$ is cointegrated with a cointegrating vector (1,-1,-1). Let $\mu = E[p_t - p_t^* - e_t]$, then μ can be nonzero when different units are used to measure prices in the two countries.

Using a one-good version of Mussa's (1982) model, the domestic price level is assumed to adjust slowly to the PPP level

$$\Delta p_{t+1} = b(\mu + p_t^* + e_t - p_t) + E_t[p_{t+1}^* + e_{t+1}] - (p_t^* + e_t)$$
(17)

¹⁰As suggested by the results in de Jong (2001), the first step estimation can affect the asymptotic distributions of the second step estimator because of the nonlinear restrictions in the system method. However, because the equations are linear in this application, the regularity conditions are likely to hold.

where $\Delta x_{t+1} = x_{t+1} - x_t$ for any variable x_t , $E[\cdot |I_t]$ is the expectation operator conditional on I_t , the information available to the economic agents at time t, and a positive constant b < 1 is the adjustment coefficient. The idea behind Equation (17) is that the domestic price level slowly adjusts toward its PPP level of $p_t^* + e_t$, while it adjusts instantaneously to the expected change in its PPP level. The adjustment speed is slow (fast) when b is close to zero (one).

From Equation (17), we obtain

$$\Delta p_{t+1} = d + b(p_t^* + e_t - p_t) + \Delta p_{t+1}^* + \Delta e_{t+1} + \varepsilon_{t+1}$$
(18)

where $d = b\mu$, $\varepsilon_{t+1} = E_t[p_{t+1}^* + e_{t+1}] - (p_{t+1}^* + e_{t+1})$. Hence ε_{t+1} is a one-period ahead forecasting error, and $E[\varepsilon_{t+1}|I_t] = 0$. Equation (18) can be referred as the structural gradual adjustment equation. In the application of this paper, the gradual adjustment equation implies the first order autoregression structure for the real exchange rate. To see this, let $s_t = p_t^* + e_t - p_t$ be the log real exchange rate. Then Equation (18) implies

$$s_{t+1} = -d + (1-b)s_t - \varepsilon_{t+1} \tag{19}$$

We define the half-life of the log real exchange rate as the number of periods required for a unit shock to dissipate by one half in this first order autoregression. Without measurement errors, the coefficient b can be estimated by Ordinary Least Squares directly from (18). In the presence of measurement error, instrumental variables are necessary. We will consider cases with and without measurement error.

4.2 The Exchange Rate under Rational Expectations

We obtain the other equations necessary for the system method estimation by adding the money demand equation and the Uncovered Interest Parity condition. Let

$$m_t = \theta_m + p_t - hi_t \tag{20}$$

$$i_t = i_t^* + E[e_{t+1}|I_t] - e_t \tag{21}$$

where m_t is the log nominal money supply minus the log real national income, i_t is the nominal interest rate in the domestic country, and i_t^* is the nominal interest rate in the foreign country. In (20), we are assuming that the income elasticity of money is one. From (20) and (21), we obtain

$$E[e_{t+1}|I_t] - e_t = (1/h)\{\theta_m + p_t - \omega_t - hE[(p_{t+1}^* - p_t^*)|I_t]\}$$
(22)

where

$$\omega_t = m_t + hr_t^* \tag{23}$$

and r_t^* is the foreign real interest rate:

$$r_t^* = i_t^* - E[p_{t+1}^*|I_t] + p_t^*$$
(24)

Following Mussa, solving (17) and (22) as a system of stochastic difference equation for $E[p_{t+j}|I_t]$ and $E[e_{t+j}|I_t]$ for fixed t results in

$$p_t = E[F_t|I_{t-1}] - \sum_{j=1}^{\infty} (1-b)^j \{ E[F_{t-j}|I_{t-j}] - E[F_{t-j}|I_{t-j-1}] \}$$
(25)

$$e_t = \frac{bh+1}{bh} E[F_t|I_t] - p_t^* - \frac{1}{bh} p_t$$
(26)

where

$$F_t = (1 - \delta) \sum_{j=0}^{\infty} \delta^j \omega_{t+j}$$
(27)

and $\delta = h/(1+h)$.

We assume that ω_t is first difference stationary. Since δ is a positive constant that is smaller than one, this implies that F_t is also first difference stationary. From (25) and (26),

$$e_t + p_t^* - p_t = \frac{bh+1}{bh} \sum_{j=1}^{\infty} (1-b)^j \{ E[F_{t-j}|I_{t-j}] - E[F_{t-j}|I_{t-j-1}] \}$$
(28)

Since the right hand side of (28) is stationary,¹¹ $e_t + p_t^* - p_t$ is stationary. Hence Equation (28) implies that (p_t, e_t, p_t^*) is cointegrated with a cointegrating vector (1,-1,-1).

4.3 Hansen and Sargent's Formula

In order to obtain a structural ECM representation from the exchange rate model, we use Hansen and Sargent's (1980, 1982) formula for linear rational expectations models. From (26), we obtain

$$\Delta e_{t+1} = \frac{bh+1}{bh} (1-\delta) E[\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | I_t] - \frac{1}{bh} \Delta p_{t+1} - \Delta p_{t+1}^* + \varepsilon_{e,t+1}$$
(29)

where $\varepsilon_{e,t+1} = \frac{bh+1}{bh} [E(F_{t+1}|I_{t+1}) - E(F_{t+1}|I_t)]$, so that the law of iterated expectation implies $E[\varepsilon_{e,t+1}|I_t] = 0$. The system method in Section 5 is applicable because this equation involves a discounted sum of expected future values of $\Delta \omega_t$.

¹¹This assumes that $E_t(F_t)$ - $E_{t-1}(F_t)$ is stationary, which is true for a large class of first difference stationary variable F_t and information sets.

Hansen and Sargent (1982) propose to project the conditional expectation of the discounted sum, $E[\delta^j \Delta y_{4,t+j+1}|I_t]$, onto an information set H_t , which is a subset of I_t , the economic agents' information set. Let $\hat{E}[\cdot | H_t]$ be the linear projection operator conditional on an information set H_t which is a subset of I_t .

We take the econometrician's information set at t, H_t , to be the one generated by the linear functions of the current and past values of Δp_t^* . Then replacing the economic agents' best forecast, $E[\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | I_t]$, by the econometrician's linear forecast based on H_t in Equation (29), we obtain

$$\Delta e_{t+1} = \frac{bh+1}{bh} (1-\delta) \widehat{E} [\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | H_t] - \frac{1}{bh} \Delta p_{t+1} - \Delta p_{t+1}^* + u_{2,t+1}$$
(30)

where

$$u_{2,t+1} = \varepsilon_{e,t+1} + \frac{bh+1}{bh} (1-\delta) E_t [(\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | I_t) - \widehat{E}(\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | H_t)]$$
(31)

Because H_t is a subset of I_t , we obtain $\widehat{E}[u_{2,t+1}|H_t] = 0$.

Since $\widehat{E}[\cdot|H_t]$ is the linear projection operator onto H_t , there exist possibly infinite order lag polynomials $\beta(L)$, $\gamma(L)$, and $\xi(L)$, such that

$$\widehat{E}[\Delta p_{t+1}^*|H_t] = \beta(L)\Delta p_t^* \tag{32}$$

$$\widehat{E}[\Delta\omega_{t+1}|H_t] = \gamma(L)\Delta p_t^*$$
(33)

$$\widehat{E}[\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1} | H_t] = \xi(L) \Delta p_t^*$$
(34)

Then following Hansen and Sargent (1980, Appendix A), we obtain the restrictions imposed by (30) on $\xi(L)$:

$$\xi(L) = \frac{\gamma(L) - \delta L^{-1} \gamma(\delta) [1 - \delta \beta(\delta)]^{-1} [1 - L\beta(L)]}{1 - \delta L^{-1}}$$
(35)

Assume that linear projections of Δp_{t+1}^* and $\Delta \omega_{t+1}$ onto H_t have only finite number of Δp_t^* terms:

$$\widehat{E}[\Delta p_{t+1}^*|H_t] = \beta_1 \Delta p_t^* + \beta_2 \Delta p_{t-1}^* + \dots + \beta_p \Delta p_{t-p+1}^*$$
(36)

$$\widehat{E}[\Delta\omega_{t+1}|H_t] = \gamma_1 \Delta p_t^* + \gamma_2 \Delta p_{t-1}^* + \dots + \gamma_{p-1} \Delta p_{t-p+2}^*$$
(37)

Here we assume $\beta(L)$ is of order p and $\gamma(L)$ is of order p-1 in order to simplify the exposition, but we do not lose generality because any of β_i and γ_i can be zero. Then as in Hansen and Sargent (1982), equation (35) implies that $\xi(L) = \xi_0 + \xi_1 L + \dots + \xi_p L^p$, where

$$\xi_0 = \gamma(\delta)[1 - \delta\beta(\delta)]^{-1}$$

$$\xi_j = \delta\gamma(\delta)[1 - \delta\beta(\delta)]^{-1}(\beta_{j+1} + \delta\beta_{j+1} + \dots + \delta^{p-j}\beta_p) + (\gamma_j + \delta\gamma_j + \dots + \delta^{p-j}\gamma_p)$$
(38)

for j = 1,...,p.

Thus

$$\widehat{E}[\sum_{j=0}^{\infty} \Delta \omega_{t+j+1} | H_t] = \xi_1 \Delta p_t^* + \xi_2 \Delta p_{t-1}^* + \dots + \xi_p \Delta p_{t-p+1}^*$$
(39)

Using (18), (30), (32), (33), and (39), we obtain a system of four equations:

$$\Delta p_{t+1} = d + \Delta p_{t+1}^* + \Delta e_{t+1} - b(p_t - p_t^* - e_t) + u_{1,t+1}$$
(40)

$$\Delta e_{t+1} = -\frac{1}{bh}\Delta p_{t+1} - \Delta p_{t+1}^* + \alpha \xi_1 \Delta p_t^* + \alpha \xi_2 \Delta p_{t-1}^* + \dots + \alpha \xi_p \Delta p_{t-p+1}^* + u_{2,t+1}$$
(41)

$$\Delta p_{t+1}^* = \beta_1 \Delta p_t^* + \beta_2 \Delta p_{t-1}^* + \dots + \beta_p \Delta p_{t-p+1}^* + u_{3,t+1}$$
(42)

$$\Delta\omega_{t+1} = \gamma_1 \Delta p_t^* + \gamma_2 \Delta p_{t-1}^* + \dots + \gamma_{p-1} \Delta p_{t-p+2}^* + u_{4,t+1}$$
(43)

where $\alpha = \frac{bh+1}{bh}(1-\delta)$ and $u_{1,t+1} = \varepsilon_{t+1}$.

Given the data for $[\Delta p_{t+1}, \Delta e_{t+1}, \Delta p_{t+1}, \Delta \omega_{t+1}]'$, GMM can be applied to these four equations as discussed in Section 3.2. There exist additional complications for obtaining data for $\Delta \omega_{t+1}$ as we discuss in Section 4.4.

The exchange rate model can be written in the SECM form (2) as in the system of equations (40)-(43): we have $\mathbf{y}_t = [\Delta p_{t+1}, \Delta e_{t+1}, \Delta p_{t+1}^*, \Delta \omega_{t+1}]', \mathbf{B} = [-b, 0, 0, 0]', \mathbf{A} = [1, -1, -1, 0]',$

$$\mathbf{C}_{0} = \begin{bmatrix} 1 & -1 & -1 & 0\\ 1/bh & 1 & 1 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{bmatrix},\tag{44}$$

and

$$\mathbf{C}_{j} = \begin{bmatrix} 0 & 0 & 0 & 0 \\ 0 & 0 & \alpha \xi_{j} & 0 \\ 0 & 0 & \beta_{j} & 0 \\ 0 & 0 & \gamma_{j} & 0 \end{bmatrix}$$
(45)

for j = 1,...,p. For any nonzero constant ψ , $\psi(1, -1, -1)'$ is also a cointegrating vector. However, the first row of **B** in (2) is b only when ψ is normalized to one.

In the exchange rate model in the previous section, b is a structural parameter of interest. For the purpose of estimating b in the model, the restriction that \mathbf{C}_0 in (2) is lower triangular is not attractive. However, as is clear from Equation (44), the structural ECM from the one-good version of the exchange rate model does not satisfy the restriction that C_0 is lower triangular for any ordering of the variables. Even though some structural models may be written in lower triangular form, this example suggests that many structural models cannot be written in that particular form.

It is instructive to observe the relationship between the structural ECM and the reduced form ECM in the exchange rate model. Because

$$\mathbf{C}_{0}^{-1} = \begin{bmatrix} \frac{bh/(bh+1) & bh/(bh+1) & 0 & 0}{-1/(bh+1) & bh/(bh+1) & -1 & 0}\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{bmatrix},$$
(46)

 $\mathbf{G} = \mathbf{C}_0^{-1}\mathbf{B} = [-b^2h/(bh+1), b/(bh+1), 0, 0]'$. Comparing **G** and **B** shows contemporaneous interactions between the domestic price and the exchange rate affect the speed of adjustment coefficients. The speed of adjustment coefficient for the domestic price is b in the structural model, while it is $b^2h/(bh+1)$ in the reduced form model. The error correction term does not appear in the second equation for the exchange rate in the structural ECM, while it appears with the speed of adjustment coefficient of b/(bh+1) in the reduced form model.

4.4 Applying the System Method to the Exchange Rate Model

In order to apply the system method to Equations (40)-(43) of the exchange rate model in the previous section, we need data for $\Delta \omega_t$, which requires the knowledge of h. Even though h is unknown, a cointegrating regression can be applied to money demand if money demand is stable in the long-run as in Stock and Watson (1993). For this purpose, we augment the model as follows:

$$m_t = \theta_m + p_t - hi_t + \zeta_{m,t} \tag{47}$$

where $\zeta_{m,t}$ is the money demand shock, which is assumed to be stationary, so that money demand is stable.

By redefining m_t as $m_t - \zeta_{m,t}$, the same equations as those in Section 4.2 are obtained. For the measurement of $\Delta \omega_t$ used in the system method, we note that the *ex ante* foreign real interest rate can be replaced by the *ex post* real foreign real interest rate because of the Law of Iterated Expectations. Using the money market clearing condition (47), we obtain

$$\Delta\omega_{t+1} = \Delta p_{t+1} - h\Delta i_{t+1} + h\Delta i_{t+1}^* - h(\Delta p_{t+2}^* - \Delta p_{t+1}^*)$$
(48)

With this expression, $\Delta \omega_t$ can be measured from price and interest rate data without data for monetary aggregate and national income once h is obtained. This is useful because the latter data are not available at the monthly frequency for many countries.

5 A Measurement Model

We apply the single equation and system methods to the exchange rate model in Section 4, using quarterly exchange rate and aggregate price data for Canada, France, Germany, Italy, Japan, the United Kingdom, and the United States from 1974:Q1 to 2001:Q1. In the model, we assume that $\mathbf{y}_t = (\Delta p_{t+1}, \Delta e_{t+1}, \Delta p_{t+1}^*)$ is cointegrated with a known cointegrating vector (1,-1,-1). This assumption may cause a problem in applications of the model to data in the post-Bretton Woods period because many researchers have failed to reject the null hypothesis of no cointegration using similar data sets. Because more favorable evidence for the assumption is often found when a longer sample period is used, the failure to reject no cointegration may be due to low power of the no cointegration tests in small samples (see, e.g., Rogoff, 1995 for a survey). Because the evidence is mixed, a sensitivity analysis with respect to this assumption is in order. For the purpose of a sensitivity analysis, we employ Cheung and Lai (1993) and Fisher and Park's (1991) model with measurement errors to allow the cointegrating vector to be different from (1,-1,-1).

5.1 Measurement Errors and the Single Equation Method

Let p_t^m and $p_t^{*,m}$ be the log measured domestic and foreign prices, which are related to the true prices by

$$p_t^m = \theta + \phi p_t + \nu_t \tag{49}$$

$$p_t^{*,m} = \theta^* + \phi^* p_t^* + \nu_t^* \tag{50}$$

where $E_{t-1}[v_t] = 0$ and $E_{t-1}[v_t^*] = 0$. We assume that true prices follow the model of Section 4 and satisfy PPP in the long-run. Then

$$p_t^m - \phi e_t - (\phi/\phi^*)p_t^* = (\theta - \theta^*\phi/\phi^*) + \phi(p_t - e_t - p_t^*) + [\nu_t - (\phi/\phi^*)\nu_t^*]$$
(51)

is stationary. Hence, $\mathbf{y}_t = (p_t^m, e_t, p_t^{*,m})'$ is cointegrated with a cointegrating vector $(1, -\phi, -\phi/\phi^*)$. In the first step, we run a cointegrating regression of the form

$$p_t^m = \psi_0 + \psi_1 e_t + \psi_2 p_t^{*,m} + \zeta_t \tag{52}$$

where $\psi_1 = \phi$, $\psi_2 = \phi/\phi^*$, and ζ_t is stationary with mean zero.

In order to obtain the second step estimator, we use Equation (18) and $\Delta p_{t+1}^m = \phi \Delta p_{t+1} + \Delta \nu_{t+1}$ to obtain

$$\Delta p_{t+1}^m = d - b[p_t^m - \phi e_t - (\phi/\phi^*)p_t^{*,m}] + (\phi/\phi^*)\Delta p_{t+1}^{*,m} + \phi\Delta e_{t+1} + w_{t+1}$$
(53)

where $d = b(\mu + \theta - \theta^* \phi / \phi^*)$, and

$$w_{t+1} = \phi \varepsilon_{t+1} + \nu_{t+1} - (1-b)\nu_t - (b\phi/\phi^*)\nu_{t+1} + (1-b)(\phi/\phi^*)\nu_t^*$$
(54)

Because $E_{t-1}[w_{t+1}] = 0$, we can apply the two step procedure from the last section as long as the instrumental variables are chosen from the information set available at t-1. In this case, the second step is to apply an IV estimation method to Equation (53), where ϕ and ϕ^* are obtained in the first step estimation. Because $E_{t-1}[w_{t+1}] = 0$ and w_{t+1} is in the information set available at t+1, w_{t+1} has a moving average (MA) representation of order one, and this serial correlation structure needs to be taken into account (see, e.g., Ogaki 1993a for an explanation of methods which treat this type of serial correlation in GMM).

5.2 Measurement Errors and the System Method

We use the measurement error model for the purpose of a sensitivity analysis with respect to PPP as in the case for the single equation method. Again it is assumed that the model is true for the true price levels, but that only measured prices that follow (49) and (50) are observed. Since p_t^m and $p_t^{*,m}$ are observed instead of p_t and p_t^* , (49) and (50) are substituted into Equations (40)-(43) in order to express these equations in terms of measured prices. It is also assumed that H_t is the information set generated by the current and past values of $\Delta p_t^{*,m}$ instead of Δp_t^* .

As for the adjustment to the PPP level, (40) is replaced by (53). For $\Delta \omega_t$, we use

$$\Delta \omega_{t+1}^m = \frac{1}{\phi} \Delta p_{t+1}^m - h \Delta i_{t+1} + h \Delta i_{t+1}^* - \frac{h}{\phi^*} (\Delta p_{t+2}^{*,m} - \Delta p_{t+1}^{*,m})$$
(55)

so that

$$\Delta e_{t+1} = d_2 + \frac{bh+1}{bh} (1-\delta) \widehat{E} [\sum_{j=0}^{\infty} \delta^j \Delta \omega_{t+j+1}^m | H_t] - \frac{1}{bh\phi} \Delta p_{t+1}^m - \frac{1}{\phi^*} \Delta p_{t+1}^{*,m} + u_{2,t+1}^m$$
(56)

where

$$u_{2,t+1}^{m} = u_{2,t+1} - \frac{bh+1}{bh} (1-\delta) \widehat{E}[\frac{1}{\phi}\nu_{t} - \frac{h}{\phi^{*}}\nu_{t}^{*}|H_{t}] + \frac{1}{bh\phi}\Delta\nu_{t+1} - \frac{1}{\phi^{*}}\Delta\nu_{t+1}^{*}$$
(57)

and $\widehat{E}[u_{2,t+1}^m | H_{t-1}] = 0.$

Because the price level is assumed to be measured with errors as in (49),

$$m_t = \theta_2 + (1/\phi)p_t^m - hi_t + \zeta_{2,t}$$
(58)

where $\theta_2 = \theta_m - \theta/\phi$ and $\zeta_{2,t} = \zeta_{m,t} - \nu_t/\phi$. Because $\zeta_{2,t}$ is stationary, a cointegrating regression is applied to (58), assuming m_t and i_t are first difference stationary.

Thus we run two cointegrating regressions, (52) and (58), in the first step. In the second step, GMM is applied to the system of four equations that consist of (53),

$$\Delta e_{t+1} = -\frac{1}{bh\phi} \Delta p_{t+1}^m - \frac{1}{\phi^*} \Delta p_{t+1}^{*,m} + \alpha \xi_1 \Delta p_t^{*,m} + \alpha \xi_2 \Delta p_{t-1}^{*,m} + \dots + \alpha \xi_p \Delta p_{t-p+1}^{*,m} + u_{2,t+1}^m$$
(59)

$$\Delta p_{t+1}^{*,m} = \beta_1 \Delta p_t^{*,m} + \beta_2 \Delta p_{t-1}^{*,m} + \dots + \beta_p \Delta p_{t-p+1}^{*,m} + u_{3,t+1}$$
(60)

$$\Delta\omega_{t+1}^{m} = \gamma_1 \Delta p_t^{*,m} + \gamma_2 \Delta p_{t-1}^{*,m} + \dots + \gamma_{p-1} \Delta p_{t-p+2}^{*,m} + u_{4,t+1}$$
(61)

where h is replaced by its estimate from (58) and ϕ and ϕ^* are replaced by their estimates from (52). As before, because the first step estimators are super consistent, the first step estimation does not affect asymptotic distributions of the second step GMM estimators under some regularity conditions.

6 Empirical Results

In this section, we present empirical results for the system methods. Quarterly foreign exchange rates and CPI from the International Financial Statistics (IFS) are used. The foreign exchange rates are stated as the domestic price of one unit of foreign currency. In this paper, we use each of the seven currencies alternatively as the base currency. The sample period is from 1974:Q1 to 2001:Q1.

For each country, we report results for two cases. The first case is when prices are measured without error, which leads to the case of the known cointegrating vector. The second case is that of the measurement error model of Section 5.1, in which the cointegrating vector for domestic prices, exchange rates and foreign prices is not restricted to be (1,-1,-1). For the latter case, the two-step method is used. In the first step, we use CCR to obtain long-run coefficients in PPP relations. In the second step, we apply GMM to estimate the short-run coefficient.

For the measurement error model, we need estimates of the coefficients in the cointegrating relationship (52), which is based on PPP. Table 1 presents the results cointegrating regressions. We report the third stage estimates of CCR for the coefficients and the fourth stage test results.

The deterministic cointegrating restrictions are rejected for 12 out of 42 cases at the five percent significance level. The restriction is not rejected for all of Italian line and Japanese year based cases at the five percent level. The null of stochastic cointegration is rejected for 14 out of 42 cases at the 5 percent level of significant. For most of U.S. dollar, German mark and Japanese year based cases, the restrictions are not rejected at the five percent significance level for any H(1,q) test in the table.

For the system method, our estimation procedure has two steps. First, we estimate the monetary equilibrium equation to obtain interest elasticity of money demand. For the measurement error model, we also obtain the measurement error coefficients, exploiting the long-run relationship between domestic prices, foreign prices and exchange rates. In the second step, the speed of price adjustment is estimated from the adjustment equation as well as the Hansen and Sargent equations.

To estimate the interest elasticity of money demand, we use the sum of M1 and Quasi Money as the measure of the money stock, called M2, as the IFS suggests. The data for interest rates are the three month T-bill rates, but three month deposit rates are employed for Japan because T-bill rates are not available. We use nominal and real gross domestic product data in the IFS dataset for all countries except the UK, for which we use the DRI data. All data series are seasonally adjusted.

Table 2 shows the CCR results for the money demand equations. We assume that the income elasticity of money demand is one. For each country, the first row reports the results when the coefficient of the log price is restricted to be one, and the second row reports the results when the coefficient is allowed to differ from one. When we employ the measurement error model, we use the results reported in the second row.

The null of stochastic cointegration is rejected only for Germany, regardless of the assumption of measurement errors at the 5 percent level. The deterministic cointegrating restriction is rejected for Germany, Italy and Japan at the 5 percent level, when we allow for measurement errors. With the prespecified cointegrating vector (1, -1, -1), France rejects the deterministic cointegrating restriction at the 5 percent level, but do not reject it at 1 percent.

In all cases, the signs of the estimates for the interest elasticity of money demand are negative,

as expected from the economic model. For Canada and France, the specification of measurement errors does not affect the estimates for the interest elasticities. However, for Germany, Italy, Japan, and the UK, the estimates from the measurement error models have smaller values than those from the models without measurement error. Interestingly, these range from one fourth to one fifth of the estimates from the no measurement error models for each country.

When we restrict the cointegrating vector to (1, -1, -1), the measurement error coefficients are no longer free parameters. In this case, we have no problem when we separately run two cointegrating regressions which include a common coefficient. But, if we allow for measurement errors in price indices, then we have two estimates for the measurement error coefficient on the domestic prices. One set of estimates is obtained from the PPP regression and the other set from the money demand equation. There is no guarantee in practice for the two estimates to be the same. If the estimates from the two equations are significantly different, it might imply misspecification of the simple exchange rate model. Although this is the case, we use the estimates from the PPP equation in Table 1 because we are more interested in PPP than in the money demand equation. Park and Ogaki (1991) suggest the seemingly unrelated canonical cointegrating regressions (SUCCR) method to deal with cross equation restrictions, when there are cointegrating vectors in the equations. However, since the small sample properties of their estimator are not better than CCR, we use the estimates from single equation CCR.

Table 3 reports the results of GMM estimation using the system method, equations (40)~(43) for the case of no measurement error, and equations (53) and (59)~(61) for the measurement error case. We also report the estimation results with additional sample period, namely 1974:Q1~1990:Q2, to see whether or not German Economic and Monetary Union affects our results. The results are not very different for the full sample and the subsample. The instrumental variables are $\Delta p_{t-3}^{*,m}$ and $\Delta p_{t-4}^{*,m}$, which are foreign prices in all cases.¹² For each country, we report results for the known cointegrating vector case and the unknown cointegrating vector case. In the system method, the structural speed of adjustment coefficient *b*, appears in two equations: the gradual adjustment equation, (18) or (40), or the Hansen-Sargent equation, (41). The model imposes the restriction that the coefficient *b* in the gradual adjustment equation is the same as the coefficient *b* in the Hansen-Sargent equation. We report results with and without this restriction imposed for the system method of estimation. In the case of unrestricted estimation, *b_{hs}* is the estimate of *b* from

¹²The selection of the instrumental variables is based on Akaike Information Criteria (AIC).

the Hansen-Sargent equation, and b_{ga} is the estimate of b from the gradual adjustment equation. The restricted estimate is denoted by b_r . The likelihood ratio type test statistic (see, e.g., Ogaki (1993a) for an explanation of this test), denoted by LR, is used to test the restriction. In all cases, this restriction is not rejected at the five percent level. Furthermore, for the test of the Hansent-Sargent restrictions in equation (38), we also report the likelihood ratio type test statistic, denoted by LR1.¹³ For all cases the null hypothesis is not rejected at the ten percent level, which is evidence in favor of the Hansen-Sargent restrictions.

To obtain half-life estimate, we use the restricted estimate of the structural speed of the adjustment coefficient, b, in each case. The half-life estimate is based on the first order autoregressive process of the domestic price implied by Equation (33). Because 1 - b is the AR coefficient for the first order AR representation, and because our data are quarterly, the half-life is calculated as $0.25 \ln(0.5)/\ln(1-b)$. All restricted estimates for the structural speed of adjustment coefficient have the theoretically correct positive sign. Most of them are significant at the five percent level. The results in Table 3 show that the half-life estimates range from 0.07 to 1.88 years. These half-life estimates are shorter than one year and much shorter than the consensus of 3-5 years when half lives are estimated by single-equation methods (see, e.g., Rogoff (1996)).

7 Concluding Remarks

This paper compares reduced form ECMs with structural form ECMs. The speed of adjustment coefficients in reduced form ECMs are different from those in structural form ECMs in general, and in our example of an exchange rate model with sticky prices. We discussed a single equation IV method and a system IV method to estimate structural speed of adjustment coefficients. These IV methods do not require exogeneity assumptions, and can be applied to a broad range of structural ECMs.

When the system method is applied to the exchange rate model, the speed of adjustment coefficient is estimated from both the gradual adjustment equation for the domestic price and the rational expectations equation for the exchange rate. The half-life estimates from the system method seem to be shorter than two years and thus are shorter than the consensus of 3-5 years when half lives are estimated by single-equation methods (see Rogoff (1996)). As explained by

¹³This test is done by conducting the likelihood ratio type test comparing the J with the Hansen-Sargent restriction from the linear rational expectations model and unrestricted one with free parameters.

Rogoff (1996), price adjustment within 1 or 2 years is plausible, and the consensus of 3-5 years is a puzzle. Given that Murray and Papell (1992) find that the single-equation methods do not give much information about the half lives, the fact that the system method yields more plausible estimates suggest that it extracts useful information about the half lives.

In this paper, we assumed the long-run PPP for CPI-based real exchange rate and uncovered interest parity (UIP) for the short-term interest rate differentials in the exchange rate model. Both of these conditions are often rejected by data. Therefore, this paper is just an initial step toward utilizing information from an economic model to estimate the structural speed of adjustment coefficient in an ECM. It is of interest to see how sensitive the results are when we relax these assumptions. Kim (2003), and Kim and Ogaki (2002) have relaxed the long-run PPP assumption by modifying the system method developed in this paper to a two-good model and applied it to traded and non-traded good prices. They find plausible results that traded good prices adjust faster than non-traded good prices.¹⁴ In the future work, we also plan to relax the UIP assumption. For example, in Lim and Ogaki's (2003) model, the UIP essentially holds for the long-term interest rate differential, but the forward premium anomaly exists for the short-term interest differential. It may be possible to develop a system method based on the UIP for the long-term interest rate differential.

¹⁴Kim (2003) follows Stockman and Tesar (1995) and uses the implicit deflators of non-service consumption and service consumption classified by type and total consumption deflators to construct the real exchange rate for traded, non-traded, and general prices, respectively. Kim finds that the estimated half-lives for the rates of the traded goods range from 0.17 to 0.91 year. For general price rates, the estimated half-lives are around 1.30 to 1.88 years and for non-traded goods' rates, the adjustment speeds to PPP are much slower than those for traded goods' prices and general prices. Kim and Ogaki (2002) use PPI, CPI, and GDP deflators to construct the real exchange rates for traded, non-traded, and general prices, respectively. They find that the estimated half-lives of the PPI-based real exchange rates range from 0.19 to 0.41 year. For the GDP deflator-based real exchange rates, the estimated half-lives range from 0.32 to 0.86 year and for the CPI-based real exchange rates the half-life estimates fall in the 0.88- to 1.57-year range. This result is consistent with Kim (2003).

REFERENCES

- Ahmed, Shaghil, Ickes, Barry W., Want, Ping, and Yoo, Byung Sam (1993), "International Business Cycles" *American Economic Review* 83, 335-59.
- (2) Boswijk, H.P. (1994), "Conditional and Structural Error Correction Models," Journal of Econometrics 69, 159-171.
- (3) _____. (1995), "Efficient Inference on Cointegration Parameters in Structural Error Correction Models," Journal of Econometrics 69, 133-158.
- (4) Campbell, J.Y. and P. Perron (1991), "Pitfalls and Opportunities: What Macroeconomists Should Know about Unit Roots" in *NBER Macroeconomics Annual* 1991, ed. by O.J. Blanchard and S. Fishcer, Cambridge, MA: MIT Press.
- (5) Campbell, J. Y., and R. J. Shiller (1987), "Cointegration and Tests of Present Value Models," Journal of Political Economy 95, 1062-1088.
- (6) Cheung, Yin-Wong and Kon S. Lai (1993), "Long-Run Purchasing Power Parity During The Recent Float," *Journal of International Economics* 34, 181-192.
- (7) Cooley, Thomas F., and Stephen F. LeRoy (1985), "A theoretical Macroeconometrics: A Critique," Journal of Monetary Economics 16, 283-308.
- (8) Cooley, T.F., and M. Ogaki (1996), "A Time Series Analysis of Real Wages, Consumption, and Asset Returns: A Cointegration-Euler Equation Approach," *Journal of Applied Econometrics* 11, 119-134.
- (9) Davidson, James E.H., David F. Hendry, Frank Srba, and Stephen Yeo (1978), "Econometric Modelling of the Aggregate Time-Series Relationship between Consumers' Expenditure and Income in the United Kingdom," *Economic Journal* 88, 661-692.
- (10) de Jong, Robert M. (2001) "Nonlinear Estimation Using Estimated Cointegrating Relations," Journal of Econometrics 101, 109-122.
- (11) Dolado, Juan, John W. Galbraith, and Anindya Banerjee (1991), "Estimating Intertemporal Quadratic Adjustment Cost Models with Integrated Series," *International Economic Review* 32, (November), 919-937.

- (12) Dornbusch, Rudiger (1976), "Expectations and Exchange Rate Dynamics," Journal of Political Economy 84, (December), 1161-76.
- (13) Edison, H. J. and B. D. Pauls, 1993, "A Re-assessment of the Relationship Between Real Exchange Rates and Real Interest Rates: 1974-1990," *Journal of Monetary Economics* 31, 165-187.
- (14) Engle, R. F. and W. J. Granger (1987), "Co-Integration and Error Correction: Representation, Estimation, and Testing," *Econometrica*, 55, 251-276.
- (15) Finn, M.G. (1989), "An Econometric Analysis of the Intertemporal General Equilibrium Approach to Exchange Rate and Current Account Determination," *Journal of International Money and Finance* 8, 467-486.
- (16) Fisher, Eric O'N and Joon Y. Park (1991), "Testing Purchasing Power Parity under the Null Hypothesis of Co-Integration," *Economic Journal* 101, 1476-1484.
- (17) Giovannini, A. and J.J. Rotemberg (1989), "Exchange-Rate Dynamics with Sticky Prices: The Deutsche Mark, 1974-1982," *Journal of Business & Economic Statistics* 7, 169-178.
- (18) Gregory, Allan W., Adrian R. Pagan, and Gregor W. Smith (1993), "Estimating Linear Quadratic Models with Integrated Processes," ed. by Peter C. B. Phillips, *Models, Methods,* and Applications of Econometrics: Essays in Honour of A.R. Bergstrom, Blackwell, Oxford, 220-39.
- (19) Hansen, Lars Peter. "Large Sample Properties of Generalized Method of Moments Estimators," *Econometrica* 50 (July 1982), 1029-1054.
- (20) Hansen, L. P., and T. J. Sargent (1980), "Formulating and Estimating Dynamic Linear Rational Expectations Models," *Journal of Economic Dynamics and Control* 2, 7-46.
- (21) _____. (1982), "Instrumental Variables Procedures for Estimating Linear Rational Expectations Models," Journal of Monetary Economics 9, 263-296.
- (22) Hsiao, Cheng (1997), "Cointegration and Dynamic Simultaneous Equations Model," *Econo*metrica 65 (3): 647-670.

- (23) Jang, K. (2000), "Impulse Response Analysis with Long Run Restrictions on Error Correction Models," manuscript, Ohio State University.
- (24) Jang, K. and M. Ogaki (2003), "The Effects of Monetary Policy Shocks on Exchange Rates: A Structural Vector Error Correction Model Approach," *Journal of the Japanese and International Economies*, forthcoming.
- (25) Johansen, S. (1988), "Statistical Analysis of Cointegrating Vectors." Journal of Economic Dynamics and Control 12, 231-254.
- (26) _____. (1991), "Estimation and Hypothesis Testing of Cointegration Vectors in Gaussian Vector Autoregressive Models," *Econometrica* 59, 1551-80.
- (27) Kilian, L., and T. Zha (2002), "Quantifying the Uncertainty about the Half-Life of Deviations from PPP," Journal of Applied Econometrics 17, 107-125.
- (28) Kim, Jaebeom (2003), "Convergence Rates to PPP for Traded and Non-traded Goods: A Structural Error Correction Model Approach," Journal of Business & Economic Statistics, forthcoming.
- (29) Kim, Jaebeom and Masao Ogaki (2002), "Purchasing Power Parity for Traded and Nontraded Goods: A Structural Error Correction Model Approach," manuscript, the Ohio State University.
- (30) King, R. G., C. I. Plosser, J. H. Stock, and M. W. Watson (1991), "Stochastic Trends and Economic Fluctuations," *American Economic Review* 81, 810-40.
- (31) Lim, Hyoung-Seok and Masao Ogaki (2003), "A Theory of Exchange Rate and The Term Structure of Interest Rates," manuscript, the Ohio State University.
- (32) Mussa, Michael (1982), "A Model of Exchange Rate Dynamics," Journal of Political Economy 90, 74-104.
- (33) Murray, C. J., and D. H. Papell (2002), "The Purchasing Power Parity Persistence Paradigm," Journal of International Economics 56, 1-19.

- (34) Ogaki, M. (1993), "Generalized Method of Moments: Econometric Applications" Handbook of Statistics, Vol.11: Econometrics, ed. by G.S. Maddala, C.R. Rao, and H.D. Vinod. Amsterdam: North-Holland, 455-488. (a)
- (35) _____. (1993) "Unit Roots in Macroeconometrics: A Survey," Monetary and Economic Studies 11, 131-154. (b)
- (36) _____. (1993) "GMM: A User's Guide," Rochester Center for Economic Research Working Paper No. 348, University of Rochester. (c)
- (37) ____. (1993) "CCR: A User's Guide," Rochester Center for Economic Research Working Paper No. 349, University of Rochester. (d)
- (38) _____. (1998) "On the Granger Representation Theorem: A Counter Example?" *Economics Letters* 60, 19-21.
- (39) Ogaki, M. and J. Y. Park, (1998), A Cointegration Approach to Estimating Preference Parameters, *Journal of Econometrics* 82, 107-134.
- (40) Pagan, A.R. and Robertson, J.C. (1995), "Structural Models of the Liquidity Effects," manuscript, Australian National University.
- (41) Papell, David (1997), "Cointegration and Exchange Rate Dynamics," Journal of International Money and Finance, 445-459.
- (42) Phillips, P.C.B. (1991), "Optimal Inference in Cointegrated Systems," *Econometrica* 59, 283-306.
- (43) Rogoff, K. (1996), "The Purchasing Power Parity Puzzle," Journal of Economic Literature 34, 647-668.
- (44) Stock, James H., and Watson, Mark W. (1993), "A Simple Estimator of Cointegrating Vectors in Higher Order Integrated Systems," *Econometrica*, 61, 783-820.
- (45) Urbain, Jean-Pierre (1992), "On Weak Exogeneity in Error Correction Models," Oxford Bulletin of Economics and Statistics 54, 187-207.

- (46) West, K. D. (1987), "A Specification Test for Speculative Bubbles," Quarterly Journal of Economics XII, 553-580.
- (47) ____. (1988), "Asymptotic Normality, When Regressors Have a Unit Root," *Econometrica*, 56, 1397-1417.
- (48) _____. (1989), "Estimation of Linear Rational Expectations Models in the Presence of Deterministic Terms," *Journal of Monetary Economics* 24, 437-442.

Table 1. Purchasing Power Parity									
$Currency^{(1)}$	$\psi_{0}^{(2)}$	$\phi^{(3)}$	$\phi/\phi^{*(4)}$	$H(0,1)^{(5)}$	$H(1,2)^{(6)}$	$H(1,3)^{(7)}$			
CA/U.S.	0.019	0.109	0.991	138.61	0.426	1.472			
	(0.244)	(0.177)	(0.063)	(0.000)	(0.513)	(0.478)			
FR/U.S.	0.006	0.237	0.906	0.401	0.231	0.921			
	(0.312)	(0.112)	(0.069)	(0.526)	(0.630)	(0.631)			
GE/U.S.	-0.053	0.195	0.957	5.237	2.436	4.357			
	(0.980)	(0.429)	(0.114)	(0.022)	(0.118)	(0.113)			
IT/U.S.	-1.045	0.185	0.910	0.488	0.327	0.472			
	(0.849)	(0.154)	(0.138)	(0.484)	(0.567)	(0.789)			
JP/U.S.	-2.469	0.282	1.252	0.858	10.121	10.512			
	(1.212)	(0.126)	(0.143)	(0.354)	(0.001)	(0.005)			
U.K./U.S.	0.438	0.067	0.899	3.054	6.566	7.201			
	(0.350)	(0.161)	(0.071)	(0.080)	(0.010)	(0.027)			
CA/U.K.	0.809	0.004	0.818	5.042	3.633	4.901			
	(0.412)	(0.245)	(0.083)	(0.024)	(0.056)	(0.086)			
FR/U.K.	-0.649	0.554	0.866	1.332	4.317	7.155			
	(0.624)	(0.211)	(0.067)	(0.248)	(0.037)	(0.027)			
GE/U.K.	2.244	-0.441	0.678	0.013	7.887	9.462			
	(0.585)	(0.152)	(0.092)	(0.907)	(0.005)	(0.008)			
IT/U.K.	-0.573	0.037	1.049	4.920	0.887	2.746			
	(0.880)	(0.136)	(0.076)	(0.026)	(0.346)	(0.253)			
JP/U.K.	5.333	-0.394	0.298	0.379	20.848	22.331			
	(1.057)	(0.101)	(0.126)	(0.539)	(0.000)	(0.000)			
U.S./U.K.	1.257	-0.290	0.749	10.663	2.448	2.479			
	(0.433)	(0.193)	(0.088)	(0.001)	(0.117)	(0.289)			
FR/CA	0131	0.171	0.911	2.821	0.152	0.205			
	(0.517)	(0.175)	(0.094)	(0.093)	(0.696)	(0.902)			
GE/CA	-0.741	0.139	0.869	0.780	7.734	9.850			
	(0.463)	(0.140)	(0.083)	(0.376)	(0.005)	(0.007)			
IT/CA	-1.198	0.164	0.988	7.251	1.278	1.438			
	(0.862)	(0.149)	(0.099)	(0.007)	(0.258)	(0.487)			
$\rm JP/CA$	0.021	0.046	0.957	0.187	13.589	17.066			
	(0.855)	(0.086)	(0.111)	(0.665)	(0.000)	(0.000)			
U.K./CA	0.689	0.042	0.833	1.978	0.502	0.711			
	(0.454)	(0.272)	(0.094)	(0.159)	(0.478)	(0.700)			
U.S./CA	0.109	0.081	0.978	1.463	5.180	5.553			
	(0.159)	(0.123)	(0.041)	(0.226)	(0.022)	(0.062)			

Table 1. Purchasing Power Parity

Table 1. Purchasing Power Parity (continued)

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$					er Parity (cor		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$Currency^{(1)}$	$\psi_0^{(2)}$	$\phi^{(3)}$	$\phi/\phi^{*(4)}$	$H(0,1)^{(5)}$	$H(1,2)^{(6)}$	$H(1,3)^{(7)}$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	CA/FR		0.134	0.988	9.934	0.329	1.840
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.271)	(0.102)	(0.048)	(0.001)	(0.565)	(0.398)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	GE/FR	-2.653	0.474	1.662	41.441	2.391	12.334
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.492)	(0.189)	(0.124)	(0.000)	(0.122)	(0.002)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	IT/FR	-2.106	0.410	0.945	0.925	4.462	4.884
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.718)	(0.181)	(0.109)	(0.336)	(0.034)	(0.086)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\rm JP/FR$	4.285	-0.407	0.341	0.039	7.699	8.181
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.864)	(0.104)	(0.125)	(0.842)	(0.005)	(0.016)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	U.K./FR	0.995	0.676	1.115	1.939	5.518	9.257
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		(0.560)	(0.198)	(0.066)	(0.163)	(0.018)	(0.009)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	U.S./FR	0.962	0.163	0.840	1.533	1.085	2.850
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.368)	(0.149)	(0.080)	(0.215)	(0.297)	(0.240)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	CA/GE	1.908	0.511	0.685	6.923	0.931	1.800
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.933)	(0.222)	(0.121)	(0.008)	(0.334)	(0.406)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	FR/GE	0.683	0.090	0.831	8.942	2.265	23.339
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.358)	(0.148)	(0.091)	(0.002)	(0.132)	(0.000)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	IT/GE	-0.254	0.032	0.982	9.989	0.150	0.587
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.640)	(0.169)	(0.206)	(0.001)	(0.698)	(0.745)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	JP/GE	0.274	0.007	0.940	0.002	8.396	8.432
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.698)	(0.096)	(0.088)	(0.956)	(0.003)	(0.014)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	U.K./GE	1.908	0.337	0.691	22.012	0.042	0.058
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.824)	(0.203)	(0.118)	(0.000)	(0.835)	(0.971)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	U.S./GE	2.382	0.920	0.724	2.780	0.045	1.473
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	_	(0.618)	(0.282)	(0.073)	(0.095)	(0.831)	(0.478)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	CA/IT	-0.123					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.757)	(0.128)	· · · ·	· · · ·	. ,	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	FR/IT	2.285	0.406	1.008	0.051	2.070	2.191
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.928)	(0.237)	(0.129)	(0.821)	(0.150)	(0.334)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	GE/IT	0.412	-0.285	0.534	1.454	15.221	15.714
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.407)	(0.086)	(0.085)	(0.227)	(0.000)	(0.000)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	JP/IT	3.217	-0.371	0.935	1.386	15.368	15.504
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		· · ·	· · · ·		(0.238)	· · · ·	(0.000)
U.S./IT 0.186 -0.220 0.604 0.172 5.958 6.337	U.K./IT	0.316	0.068	0.815	1.568	0.002	5.814
•		· · ·	. ,	· · · ·	. ,	(0.963)	
(0.567) (0.097) (0.079) (0.678) (0.014) (0.042)	U.S./IT	0.186	-0.220	0.604	0.172	5.958	6.337
		(0.567)	(0.097)	(0.079)	(0.678)	(0.014)	(0.042)

Table 1.	Purchasing	Power Parity	(continued)

				- (/	
$Currency^{(1)}$	$\psi_{0}^{(2)}$	$\phi^{(3)}$	$\phi/\phi^{*(4)}$	$H(0,1)^{(5)}$	$H(1,2)^{(6)}$	$H(1,3)^{(7)}$
CA/JP	2.952	0.278	0.619	0.169	0.004	6.004
	(0.669)	(0.070)	(0.085)	(0.680)	(0.947)	(0.050)
FR/JP	2.837	0.268	0.542	0.118	0.901	7.924
	(0.813)	(0.102)	(0.117)	(0.730)	(0.342)	(0.019)
GE/JP	2.082	0.249	0.732	3.126	0.665	0.711
	(0.617)	(0.092)	(0.072)	(0.077)	(0.414)	(0.700)
IT/JP	2.925	0.414	0.098	2.590	0.185	8.014
	(0.596)	(0.100)	(0.172)	(0.107)	(0.666)	(0.018)
U.K./JP	2.580	0.212	0.660	0.731	1.562	6.225
	(1.242)	(0.120)	(0.146)	(0.392)	(0.211)	(0.044)
U.S./JP	3.972	0.369	0.538	0.389	0.254	1.553
	(0.692)	(0.080)	(0.078)	(0.532)	(0.614)	(0.459)

Note: Results for $p_t^m = \psi_0 + \phi e_t + (\phi/\phi^*) p_t^{*,m} + \zeta_t$

Column (1): currencies

Column (2) \sim (4): Standard errors are in parentheses.

Column (5) \sim (7): P-values are in parentheses.

	Table 2. Money Demand Equation								
$Country^{(1)}$	$ heta_2^{(2)}$	$1/\phi^{(3)}$	$h^{(4)}$	$H(0,1)^{(5)}$	$H(1,2)^{(6)}$	$H(1,3)^{(7)}$			
CA	-0.046	1	30.031	1.019	0.445	1.098			
	(0.247)		(9.875)	(0.313)	(0.505)	(0.578)			
	0.464	1.899	40.791	0.523	4.681	4.792			
	(0.456)	(0.240)	(16.200)	(0.469)	(0.030)	(0.091)			
FR	-0.341	1	5.661	4.823	0.963	1.183			
	(0.071)		(3.182)	(0.028)	(0.326)	(0.554)			
	-0.337	0.253	5.560	2.036	0.097	0.321			
	(0.014)	(0.038)	(0.636)	(0.153)	(0.755)	(0.851)			
GE	-0.272	1	17.882	1.537	1.600	4.575			
	(0.165)		(9.856)	(0.215)	(0.206)	(0.102)			
	-0.454	1.597	3.003	11.500	18.610	18.650			
	(0.022)	(0.050)	(1.307)	(0.001)	(0.000)	(0.000)			
IT	-0.205	1	7.847	3.755	0.753	2.424			
	(0.172)		(4.807)	(0.053)	(0.386)	(0.298)			
	-0.508	0.078	1.994	9.247	1.560	5.979			
	(0.041)	(0.013)	(1.073)	(0.002)	(0.212)	(0.050)			
JP	-11.312	1	39.661	3.274	3.229	5.353			
	(0.059)		(5.827)	(0.070)	(0.072)	(0.069)			
	-13.951	1.520	8.089	13.020	0.035	0.674			
	(0.784)	(0.171)	(5.228)	(0.000)	(0.852)	(0.714)			
U.K.	10.662	1	111.312	1.853	0.088	0.597			
	(3.410)		(128.3)	(0.173)	(0.766)	(0.742)			
	8.982	2.265	27.560	2.942	0.129	5.562			
	(0.273)	(0.116)	(9.306)	(0.086)	(0.720)	(0.062)			
U.S.	-5.151	1	12.779	0.001	1.678	17.232			
	(0.156)		(6.875)	(0.965)	(0.195)	(0.000)			
	-4.951	0.960	14.208	1.631	2.735	24.362			
	(0.781)	(0.168)	(7.507)	(0.201)	(0.098)	(0.000)			

Table 2. Money Demand Equation

Note: Results for $m_t = \theta_2 + (1/\phi)p_t^m - hi_t + \zeta_{2,t}$

Column (1): domestic countries

Column (2)~(4): Standard errors are in parentheses.

Column (5)~(7): P-values are in parentheses.

Table 3.	The system	Method	Results for	CPI-based	Real	Exchange Rates
----------	------------	--------	-------------	-----------	------	----------------

Country ⁽¹⁾	$\phi^{(2)}$	$\phi/\phi^{*(3)}$	$\frac{\text{stem Method}}{\text{Half-Life}^{(4)}}$	$b_r^{(5)}$	$J_r^{(6)}$	$b_{u,hs}^{(7)}$	$b_{u,ga}^{(8)}$	$J_{u}^{(9)}$	$LR^{(10)}$	$LR1^{(11)}$
CA/U.S.	1	1	0.19	0.593	4.511	781.45	0.144	4.286	0.225	0.952
,			(0.039)	(0166)	(0.341)	(17566)	(0.646)	(0.232)	(0.635)	(0.621)
	0.109	0.991	0.17	0.634	8.272	2344.2	0.318	7.716	0.556	3.601
			(0.027)	(0.164)	(0.082)	(52805)	(2.678)	(0.052)	(0.455)	(0.165)
FR/U.S.	1	1	1.51	0.108	2.594	0.204	0.099	1.692	0.902	1.834
			(1.276)	(0.011)	(0.627)	(0.099)	(0.135)	(0.638)	(0.342)	(0.399)
	0.237	0.906	0.90	0.174	2.490	0.324	1.868	2.046	0.444	0.688
			(0.322)	(0.013)	(0.646)	(0.134)	(0.500)	(0.562)	(0.505)	(0.709)
GE/U.S.	1	1	0.35	0.387	6.806	0.345	0.023	3.322	3.484	1.735
			(0.094)	(0.064)	(0.146)	(0.002)	(0.007)	(0.344)	(0.062)	(0.420)
$73{:}\mathrm{I}{\sim}90{:}\mathrm{II}$			0.29	0.449	4.623	2677.8	-0.456	4.395	0.228	2.728
			(0.134)	(0.164)	(0.328)	(30464)	(0.736)	(0.221)	(0.633)	(0.255)
	0.195	0.957	0.19	0.605	1.798	0.074	0.781	1.117	0.681	0.207
			(0.067)	(0.312)	(0.772)	(0.065)	(0.341)	(0.772)	(0.409)	(0.901)
73:I~90:II			0.18	0.619	4.926	3874.9	0.556	4.517	0.409	3.538
			(0.055)	(0.288)	(0.294)	(77527)	(0.374)	(0.210)	(0.522)	(0.170)
IT/U.S.	1	1	0.26	0.489	2.618	2698.8	0.754	2.054	0.564	1.554
			(0.154)	(0.270)	(0.623)	(27157)	(1.250)	(0.561)	(0.452)	(0.459)
	0.185	0.901	0.10	0.816	1.598	0.971	0.361	1.385	0.213	0.237
			(0.014)	(0.402)	(0.808)	(2.270)	(0.239)	(0.708)	(0.644)	(0.888)
JP/U.S.	1	1	0.43	0.330	4.556	601.31	7.688	1.876	2.680	2.222
			(0.264)	(0.098)	(0.335)	(61672)	(4.289)	(0.598)	(0.101)	(0.329)
	0.282	1.252	0.23	0.524	2.828	0.130	0.803	1.789	1.039	0.491
			(0.093)	(0.221)	(0.586)	(0.021)	(0.663)	(0.617)	(0.308)	(0.782)
U.K./U.S.	1	1	0.17	0.628	2.750	0.063	3.803	2.548	0.202	2.548
			(0.026)	(0.146)	(0.600)	(0.021)	(2.874)	(0.466)	(0.653)	(0.279)
	0.067	0.899	0.72	0.213	5.621	938.78	12.350	5.479	0.142	4.273
			(1.475)	(0.117)	(0.229)	(21638)	(4.049)	(0.139)	(0.706)	(0.118)
CA/U.K.	1	1	0.47	0.308	1.685	0.024	0.816	0.206	1.479	0.413
			(0.013)	(0.004)	(0.793)	(0.000)	(0.891)	(0.976)	(0.223)	(0.813)
	0.004	0.818	0.25	0.497	4.144	0.042	2.499	1.435	2.709	2.817
			(0.263)	(0.493)	(0.386)	(0.020)	(0.564)	(0.697)	(0.099)	(0.244)
GE/U.K.	1	1	0.12	0.761	3.953	0.507	3.064	3.623	0.330	0.663
FOT 00 T			(0.009)	(0.153)	(0.412)	(0.265)	(1.019)	(0.305)	(0.565)	(0.717)
73:I~90:II			0.07	0.905	6.131	0.284	1.065	5.970	0.161	0.382
	0 4 4 7	0.070	(0.002)	(0.183)	(0.189)	(0.140)	(0.201)	(0.113)	(0.688)	(0.826)
	-0.441	0.678	0.37	0.371	2.937	0.332	0.014	0.649	2.288	2.669
			(0.740)	(0.426)	(0.568)	(0.001)	(0.028)	(0.884)	(0.130)	(0.263)
73:I~90:II			0.33	0.404	3.818	6669.8	3.396	2.946	0.872	0.115
			(0.177)	(0.142)	(0.431)	(91526)	(0.828)	(0.399)	(0.350)	(0.944)

Table 3. The system Method Results for CPI-based Real Exchange Rates (continued)	
--	--

Country ⁽¹⁾	$\phi^{(2)}$	$\phi/\phi^{*(3)}$	Half-Life ⁽⁴⁾	$b_r^{(5)}$	$J_{r}^{(6)}$	$b_{u,hs}^{(7)}$	$b_{u,ga}^{(8)}$	$J_{u}^{(9)}$	$LR^{(10)}$	$LR1^{(11)}$
FR/U.K.	1	1	0.15	0.677	2.701	0.215	2.011	1.851	0.850	1.038
			(0.035)	(0.295)	(0.609)	(0.121)	(0.596)	(0.603)	(0.356)	(0.595)
	0.554	0.866	0.12	0.768	1.640	889.60	0.776	1.578	0.062	1.378
			(0.021)	(0.390)	(0.801)	(42899)	(0.379)	(0.664)	(0.803)	(0.502)
IT/U.K.	1	1	0.22	0.545	6.808	0.330	1.369	6.611	0.197	4.450
			(0.027)	(0.078)	(0.146)	(0.363)	(0.078)	(0.085)	(0.657)	(0.108)
	0.037	1.049	0.14	0.718	1.179	0.507	0.075	1.085	0.091	0.055
			(0.039)	(0.457)	(0.881)	(0.023)	(0.110)	(0.780)	(0.762)	(0.972)
JP/U.K.	1	1	0.13	0.745	2.519	3281.6	0.997	2.060	0.459	0.137
			(0.025)	(0.380)	(0.641)	(63132)	(0.806)	(0.559)	(0.498)	(0.933)
	-0.394	0.298	0.09	0.842	4.878	0.066	0.677	3.933	0.945	1.882
			(0.005)	(0.187)	(0.300)	(0.087)	(0.213)	(0.268)	(0.331)	(0.390)
U.S./U.K.	1	1	0.59	0.253	4.223	1718.8	4.682	1.519	2.704	3.055
			(0.244)	(0.035)	(0.376)	(15995)	(2.931)	(0.677)	(0.100)	(0.217)
	-0.290	0.749	0.08	0.873	3.112	0.283	0.963	3.085	0.027	0.331
			(0.003)	(0.162)	(0.539)	(1.025)	(0.163)	(0.378)	(0.869)	(0.847)
$\mathrm{FR/CA}$	1	1	2.22	0.075	4.344	0.144	2.382	2.665	1.679	0.911
			(43.511)	(0.119)	(0.361)	(0.058)	(0.751)	(0.446)	(0.195)	(0.634)
	0.171	0.911	0.14	0.703	5.483	0.362	-1.466	4.521	0.962	2.058
			(0.033)	(0.348)	(0.241)	(0.643)	(0.375)	(0.210)	(0.326)	(0.357)
GE/CA	1	1	0.66	0.230	1.957	0.327	2.091	1.788	0.169	0.271
			(0.397)	(0.041)	(0.743)	(0.116)	(0.263)	(0.617)	(0.681)	(0.873)
$73:I\sim90:II$			0.27	0.471	5.384	0.369	0.608	4.992	0.392	0.859
			(0.118)	(0.176)	(0.250)	(0.762)	(0.296)	(0.172)	(0.531)	(0.650)
	0.139	0.869	0.17	0.629	4.701	8748.3	0.045	4.477	0.224	1.312
			(0.046)	(0.264)	(0.319)	(90024)	(0.180)	(0.214)	(0.636)	(0.519)
73:I~90:II			0.18	0.615	4.955	7113.2	0.502	4.667	0.288	0.634
			(0.067)	(0.340)	(0.291)	(2292.8)	(0.361)	(0.197)	(0.591)	(0.728)
IT/CA	1	1	0.89	0.177	4.932	3062.5	-0.398	4.344	0.588	1.884
	0.040	0.055	(11.017)	(0.470)	(0.294)	(44341)	(0.664)	(0.226)	(0.443)	(0.389)
	0.046	0.957	1.15	0.140	2.370	0.281	4.023	2.335	0.035	3.126
	1	1	(1.464)	(0.029)	(0.667)	(0.068)	(2.778)	(0.505)	(0.851)	(0.209)
JP/CA	1	1	0.30	0.437	5.401	2207.1	0.233	1.914	3.487	0.591
	0.040	0.057	(0.552)	(0.605)	(0.248)	(33989)	(3.500)	(0.590)	(0.061)	(0.744)
	0.046	0.957	1.15	0.140	2.370	0.281	4.023	2.335	0.035	0.086
	1	1	(1.464)	(0.029)	(0.667)	(0.068)	(2.778)	(0.505)	(0.851)	(0.958)
U.K./CA	1	1	0.13	0.733	3.123	0.022	0.718	3.109	0.014	1.393
	0.049	0 099	(0.013)	(0.180)	(0.537)	(0.016)	(0.639)	(0.375)	(0.905)	(0.498)
	0.042	0.833	0.42	0.341	3.396	-0.076	1.044	1.724	1.672	1.545
			(0.403)	(0.169)	(0.493)	(0.070)	(0.112)	(0.631)	(0.195)	(0.461)

Table 3. The system M	ethod Results for CPI-based Real	Exchange Rates	(continued)
-----------------------	----------------------------------	----------------	-------------

Country ⁽¹⁾	$\frac{\phi^{(2)}}{\phi^{(2)}}$	$\phi/\phi^{*(3)}$	Half-Life ⁽⁴⁾	$\frac{b_r^{(5)}}{b_r^{(5)}}$	$\frac{J_r^{(6)}}{J_r^{(6)}}$	$b_{u,hs}^{(7)}$	$\frac{b_{u,ga}^{(8)}}{b_{u,ga}^{(8)}}$	$\frac{J_u^{(9)}}{J_u^{(9)}}$	$LR^{(10)}$	LR1 ⁽¹¹⁾
U.S./CA	$\frac{\psi}{1}$	$\frac{\psi}{\psi}$	0.36	$\frac{v_r}{0.386}$	$\frac{J_r}{3.854}$	$\frac{v_{u,hs}}{-0.047}$	$\frac{v_{u,ga}}{0.814}$	$\frac{J_u}{1.729}$	2.125	0.477
0.5./CA	1	1								
	0.001	0.079	(0.189)	(0.127)	(0.425)	(0.039)	(0.201)	(0.630)	(0.144)	(0.787)
	0.081	0.978	0.08	0.878	4.992	0.044	0.857	1.960 (0.580)	3.032	1.618
	1	1	(0.003) 0.59	(0.202) 0.256	(0.288) 5.749	(0.043) 0.042	(0.182) 0.949	(0.380) 5.430	(0.081) 0.319	(0.445) 0.490
CA/FR	1	1	(0.268)		(0.218)				(0.519) (0.572)	(0.490) (0.782)
	0 194	0.988	(0.208) 0.27	$(0.040) \\ 0.468$	(0.218) 6.471	(0.004) 913.77	$(0.131) \\ 0.699$	(0.142) 5.308	(0.572) 1.163	()
	0.134	0.966	(0.150)	(0.218)		(65601)	(1.003)	(0.105)	(0.280)	1.201
CE/ED	1	1	1.07	· · · ·	(0.166)	0.300	(1.003) 0.288	(0.103) 1.868	0.003	(0.548) 0.202
GE/FR	1	1		0.149	1.871					
73:I~90:II			$(1.031) \\ 0.88$	(0.025)	(0.759)	$(0.102) \\ 0.368$	$(0.334) \\ 0.841$	$(0.600) \\ 0.580$	$(0.956) \\ 2.378$	(0.903)
75:1~90:11			(1.196)	0.178	2.958	(0.025)	(0.232)	(0.901)	(0.123)	1.086
	0.474	1.662	(1.190) 0.26	$(0.052) \\ 0.482$	$(0.564) \\ 3.600$	(0.023) 0.001	(0.232) 0.693	(0.901) 2.823	(0.123) 0.777	$(0.581) \\ 1.096$
	0.474	1.002	(0.134)		(0.462)	(0.001)			(0.378)	(0.578)
73:I~90:II			(0.134) 0.30	$(0.221) \\ 0.441$	(0.402) 7.357	(0.000) 0.175	(0.191) 1.414	$(0.419) \\ 6.622$	(0.378) 0.735	(0.578) 1.111
75.1 ~ 90.11			(0.133)	(0.151)	(0.118)	(2.587)	(0.264)	(0.022)	(0.391)	(0.573)
IT/FR	1	1	0.21	(0.131) 0.564	$\frac{(0.118)}{2.059}$	$\frac{(2.587)}{0.526}$	(0.204) 0.371	(0.034) 1.997	(0.391) 0.062	$\frac{(0.373)}{0.748}$
$11/\Gamma$ K	1	1	(0.107)	(0.356)	(0.724)	(0.076)	(1.129)	(0.572)	(0.803)	(0.000)
	0.410	0.945	0.44	(0.330) 0.325	(0.724) 2.508	(0.070) 1519.6	(1.129) -0.285	(0.572) 1.641	(0.803) 0.867	(0.000) 0.774
	0.410	0.340	(0.299)	(0.105)	(0.643)	(19070)	(1.406)	(0.650)	(0.351)	(0.688)
JP/FR	1	1	1.04	0.154	3.459	2896.1	2.005	(0.050) 2.570	0.889	0.357
51/110	1	T	(14.856)	(0.401)	(0.483)	(60717)	(1.110)	(0.462)	(0.345)	(0.836)
	-0.407	0.341	0.15	(0.101) 0.677	5.023	0.247	1.129	(0.102) 2.799	2.224	(0.050) 1.469
	0.101	0.011	(0.046)	(0.384)	(0.284)	(0.164)	(0.204)	(0.423)	(0.135)	(0.479)
U.K./FR	1	1	0.51	0.288	2.210	0.307	0.500	1.451	0.759	0.071
0.111./110	1	1	(0.168)	(0.038)	(0.697)	(6.941)	(0.538)	(0.693)	(0.383)	(0.965)
	0.676	1.115	0.43	0.331	5.589	896.65	2.094	4.347	1.242	0.235
			(0.376)	(0.141)	(0.231)	(72470)	(1.009)	(0.226)	(0.265)	(0.889)
U.S./FR	1	1	0.48	0.301	5.957	4817.9	1.311	4.627	1.330	0.586
			(0.430)	(0.114)	(0.202)	(99270)	(0.190)	(0.201)	(0.248)	(0.746)
	0.163	0.840	0.28	0.457	2.797	0.036	0.787	2.620	0.177	0.249
			(0.153)	(0.202)	(0.592)	(0.037)	(0.211)	(0.453)	(0.673)	(0.882)
CA/IT	1	1	0.13	0.739	2.877	0.209	0.988	1.433	1.444	0.163
- /	-	-	(0.017)	(0.246)	(0.578)	(38.050)	(0.209)	(0.697)	(0.229)	(0.921)
	-0.236	0.664	1.63	0.101	1.621	9.656	1.527	1.429	0.192	0.606
			(7.609)	(0.053)	(0.804)	(28.380)	(0.599)	(0.698)	(0.661)	(0.738)
FR/IT	1	1	0.20	0.577	3.204	1556.1	0.924	1.101	1.923	2.428
-/	-	_	(0.053)	(0.198)	(0.053)	(77039)	(0.645)	(0.776)	(0.165)	(0.297)
	0.406	1.008	0.12	0.777	6.831	0.413	0.999	5.035	1.796	2.144
			(0.101)	(0.203)	(0.145)	(0.135)	(0.265)	(0.169)	(0.180)	(0.342)
			()	()	()	()	()	()	()	()

Table 3. The system Method Results for CPI-based Real Exchange Rates (cor	ontinued)
---	-----------

$Country^{(1)}$	$\phi^{(2)}$	$\phi/\phi^{*(3)}$	Half-Life ^{(4)}	$b_r^{(5)}$	$J_{r}^{(6)}$	$b_{u,hs}^{(7)}$	$b_{u,ga}^{(8)}$	$\frac{\text{continued}}{J_u^{(9)}}$	$LR^{(10)}$	$LR1^{(11)}$
JP/IT	1	1	1.10	0.146	0.244	1047.5	0.263	0.208	0.036	2.053
,			(10.182)	(0.231)	(0.993)	(40154)	(0.634)	(0.976)	(0.849)	(0.358)
	-0.371	0.935	1.88	0.088	0.864	0.123	-0.062	0.457	0.407	0.659
			(95.997)	(0.433)	(0.929)	(0.01)	(0.282)	(0.928)	(0.523)	(0.719)
U.K./IT	1	1	0.14	0.708	3.283	1029.7	0.887	2.241	1.042	1.050
,			(0.016)	(0.175)	(0.511)	(67449)	(0.261)	(0.523)	(0.307)	(0.591)
	0.068	0.815	0.09	0.849	3.004	269.83	0.859	2.850	0.154	0.890
			(0.005)	(0.224)	(0.557)	(66145)	(0.225)	(0.415)	(0.694)	(0.640)
U.S./IT	1	1	0.24	0.512	1.307	0.098	0.905	0.805	0.502	0.088
,			(0.368)	(0.785)	(0.904)	(0.089)	(0.626)	(0.848)	(0.478)	(0.956)
	-0.220	0.604	0.44	0.325	4.699	0.073	0.575	3.854	0.845	0.435
			(0.376)	(0.132)	(0.319)	(0.067)	(0.396)	(0.277)	(0.357)	(0.804)
GE/IT	1	1	0.16	0.669	3.762	0.031	0.883	1.801	1.961	1.788
			(0.027)	(0.215)	(0.439)	(0.109)	(0.457)	(0.614)	(0.161)	(0.409)
$73{:}\mathrm{I}{\sim}90{:}\mathrm{II}$			0.13	0.727	4.762	0.603	1.352	2.435	2.327	0.505
			(0.011)	(0.147)	(0.312)	(0.173)	(0.672)	(0.487)	(0.127)	(0.776)
	-0.285	0.534	0.25	0.503	2.727	0.112	-1.617	2.711	0.016	0.467
			(0.120)	(0.238)	(0.604)	(0.047)	(5.152)	(0.438)	(0.899)	(0.791)
$73{:}\mathrm{I}{\sim}90{:}\mathrm{II}$			0.24	0.513	2.287	0.176	1.647	0.572	1.715	0.526
			(0.133)	(0.288)	(0.683)	(0.164)	(1.341)	(0.902)	(0.190)	(0.768)
CA/JP	1	1	0.36	0.381	6.555	904.17	1.151	4.009	2.546	1.201
			(0.252)	(0.161)	(0.161)	(52946)	(0.387)	(0.260)	(0.110)	(0.548)
	0.278	0.619	1.86	0.089	5.005	0.024	0.234	3.093	1.912	1.489
			(6.849)	(0.032)	(0.286)	(0.003)	(0.058)	(0.377)	(0.166)	(0.475)
FR/JP	1	1	0.69	0.221	4.447	-0.125	0.655	2.408	2.039	1.517
			(1.223)	(0.110)	(0.348)	(20.860)	(0.781)	(2.492)	(0.153)	(0.468)
	0.268	0.542	0.37	0.372	1.395	0.016	1.042	1.633	0.762	0.317
			(0.261)	(0.152)	(0.844)	(0.052)	(0.065)	(0.888)	(0.382)	(0.853)
GE/JP	1	1	0.51	0.286	3.032	5.617	1.672	2.127	0.905	0.899
			(1.790)	(0.395)	(0.552)	(6.422)	(0.502)	(0.546)	(0.341)	(0.638)
$73:I\sim90:II$			0.49	0.297	0.881	0.324	0.479	0.357	0.524	0.354
			(0.483)	(0.122)	(0.927)	(0.011)	(0.870)	(0.948)	(0.469)	(0.837)
	0.249	0.732	0.92	0.172	0.842	0.024	0.576	0.391	0.451	0.580
			(2.216)	(0.086)	(0.932)	(0.074)	(0.732)	(0.942)	(0.501)	(0.748)
73:I~90:II			0.86	0.182	0.389	-0.005	0.050	0.347	0.042	0.015
			(1.880)	(0.088)	(0.983)	(1.266)	(0.880)	(0.951)	(0.837)	(0.992)
IT/JP	1	1	1.42	0.115	0.868	0.501	0.709	0.103	0.765	0.029
	0.11.1	0.000	(10.264)	(0.108)	(0.928)	(0.001)	(0.235)	(0.991)	(0.381)	(0.985)
	0.414	0.098	0.36	0.385	6.049	2839.2	2.961	4.050	1.999	2.050
			(0.092)	(0.061)	(0.195)	(83321)	(0.963)	(0.256)	(0.157)	(0.358)

Table 3. The system Method Results for CPI-based Real Exchange Rates (continued)	
--	--

Country ⁽¹⁾	$\frac{1}{\phi^{(2)}}$	$\phi/\phi^{*(3)}$	Half-Life ⁽⁴⁾	$b_{r}^{(5)}$	$J_{r}^{(6)}$	$b_{u,hs}^{(7)}$	$b_{u,ga}^{(8)}$	$J_{u}^{(9)}$	$LR^{(10)}$	$LR1^{(11)}$
U.K./JP	1	1	0.66	0.232	2.205	0.042	0.874	0.731	1.474	0.121
			(0.207)	(0.022)	(0.697)	(0.012)	(0.273)	(0.865)	(0.224)	(0.941)
	0.212	0.660	0.17	0.646	5.547	926.78	0.939	3.914	1.633	1.826
			(0.031)	(0.202)	(0.235)	(80478)	(0.411)	(0.270)	(0.201)	(0.401)
U.S./JP	1	1	0.32	0.414	3.412	-0.066	-5.061	1.006	2.415	0.683
			(0.489)	(0.431)	(0.489)	(0.021)	(9.738)	(0.799)	(0.120)	(0.710)
	0.369	0.538	0.11	0.802	2.684	1422.8	2.671	2.589	0.095	0.353
			(0.012)	(0.297)	(0.611)	(11120)	(1.052)	(0.459)	(0.757)	(0.838)
CA/GE	1	1	0.12	0.754	5.797	0.128	1.456	4.911	0.886	1.220
			(0.020)	(0.326)	(0.214)	(0.075)	(0.213)	(0.178)	(0.346)	(0.543)
$73:I\sim90:II$			0.10	0.821	4.545	411.48	3.068	1.936	2.609	0.465
			(0.004)	(0.133)	(0.337)	(11620)	(7.583)	(0.585)	(0.106)	(0.792)
	0.511	0.685	0.23	0.530	5.281	0.397	1.639	5.204	0.077	0.749
			(0.037)	(0.092)	(0.259)	(1.510)	(0.169)	(0.157)	(0.781)	(0.687)
$73{:}\mathrm{I}{\sim}90{:}\mathrm{II}$			0.23	0.523	6.680	0.112	1.425	5.725	0.955	2.571
			(0.059)	(0.140)	(0.153)	(0.061)	(0.246)	(0.125)	(0.328)	(0.276)
FR/GE	1	1	0.35	0.393	7.691	0.016	1.549	6.246	1.445	3.470
			(0.203)	(0.146)	(0.103)	(0.360)	(0.461)	(0.100)	(0.229)	(0.176)
73:I~90:II			0.22	0.545	0.865	0.179	0.005	0.448	0.417	0.020
			(0.057)	(0.161)	(0.929)	(0.004)	(0.071)	(0.930)	(0.518)	(0.990)
	0.090	0.831	0.50	0.295	7.054	0.111	0.875	6.258	0.796	0.838
			(0.503)	(0.124)	(0.133)	(0.067)	(1.289)	(0.099)	(0.372)	(0.657)
$73:I \sim 90:II$			0.37	0.371	7.616	0.033	1.484	6.206	1.410	1.314
			(0.168)	(0.097)	(0.106)	(3.177)	(1.524)	(0.102)	(0.235)	(0.518)
IT/GE	1	1	0.33	0.410	0.649	-0.576	-1.872	0.272	0.377	0.049
			(0.265)	(0.225)	(0.957)	(0.196)	(5.321)	(0.965)	(0.539)	(0.975)
$73:I \sim 90:II$			0.12	0.776	1.089	0.605	0.151	0.597	0.492	0.504
			(0.014)	(0.281)	(0.895)	(0.154)	(0.054)	(0.288)	(0.483)	(0.777)
	0.032	0.982	0.47	0.309	2.201	0.525	-0.027	1.953	0.248	1.256
			(0.264)	(0.077)	(0.698)	(0.012)	(0.030)	(0.582)	(0.618)	(0.533)
$73:I \sim 90:II$			0.49	0.296	2.261	0.026	6.196	1.959	0.302	1.520
			(0.288)	(0.072)	(0.687)	(0.378)	(11.340)	(0.581)	(0.582)	(0.467)
JP/GE	1	1	0.42	0.337	1.551	-0.091	-333.34	0.590	0.961	0.282
HO T 00 T			(0.084)	(0.034)	(0.817)	(0.171)	(113.6)	(0.898)	(0.326)	(0.868)
73:I~90:II			0.39	0.357	4.930	0.169	0.433	2.516	2.414	0.884
	0.007	0.040	(0.247)	(0.123)	(0.294)	(0.048)	(0.112)	(0.472)	(0.120)	(0.642)
	0.007	0.940	1.46	0.112	4.775	0.104	1.860	2.550	2.225	1.307
			(3.515)	(0.034)	(0.311)	(0.059)	(0.551)	(0.466)	(0.135)	(0.520)
73:I~90:II			0.86	0.182	2.951	0.131	0.225	0.653	2.298	1.885
			(0.662)	(0.031)	(0.566)	(0.010)	(0.114)	(0.253)	(0.129)	(0.389)

$\frac{1}{(1)}$										
$Country^{(1)}$	$\phi^{(2)}$	$\phi/\phi^{*(3)}$	$\text{Half-Life}^{(4)}$	$b_r^{(5)}$	$J_r^{(6)}$	$b_{u,hs}^{(7)}$	$b_{u,ga}^{(8)}$	$J_u^{(9)}$	$LR^{(10)}$	$LR1^{(11)}$
U.K./GE	1	1	0.14	0.721	1.826	0.036	0.077	1.372	0.454	0.757
			(0.014)	(0.172)	(0.767)	(0.000)	(0.302)	(0.711)	(0.500)	(0.684)
$73:I\sim90:II$			0.11	0.798	6.576	1322.1	2.397	3.791	2.785	0.456
			(0.010)	(0.250)	(0.160)	(65054)	(2.019)	(0.284)	(0.095)	(0.796)
	0.337	0.691	0.18	0.628	3.172	-0.046	1.303	2.604	0.568	0.059
			(0.026)	(0.146)	(0.529)	(0.013)	(0.766)	(0.456)	(0.451)	(0.971)
$73:I\sim90:II$			0.18	0.629	3.104	0.036	0.196	1.894	1.210	1.108
			(0.047)	(0.267)	(0.540)	(0.001)	(0.161)	(0.594)	(0.271)	(0.574)
U.S./GE	1	1	0.34	0.398	3.393	1549.7	1.079	3.065	0.328	0.108
			(0.027)	(0.021)	(0.494)	(41562)	(0.297)	(0.381)	(0.566)	(0.947)
$73:I\sim90:II$			0.35	0.389	5.832	0.063	0.772	2.960	2.872	2.531
			(1.059)	(0.731)	(0.211)	(0.031)	(0.561)	(0.397)	(0.090)	(0.282)
	0.920	0.724	0.94	0.169	0.427	1117.9	0.125	0.263	0.164	0.073
			(0.874)	(0.032)	(0.980)	(42207)	(0.401)	(0.966)	(0.685)	(0.964)
$73{:}\mathrm{I}{\sim}90{:}\mathrm{II}$			0.16	0.653	5.201	917.37	0.827	5.051	0.150	0.734
			(0.033)	(0.229)	(0.267)	(43147)	(0.051)	(0.168)	(0.698)	(0.692)

Table 3. The system Method Results for CPI-based Real Exchange Rates (continued)

Note; For the unresticted estimation, $b_{u,hs}$ is the estimate for the speed of adjustment coefficient obtained from Hansen and Sargent equations, and $b_{u,ga}$ is the estimate for the coefficient obtained from the price adjustment equation. LR1 is the results of likelihood ratio type tests for the hansen-sargent restriction.

Column (1): currencies

Column (2) and (3) are from Table 1.

Column (4), (5), (7) & (8): Standard errors are in parentheses.

Column (6), (9) , (10) & (11): P-values are in parentheses.

Column (4): Half-life in years.